

Received March 12, 2017, accepted April 2, 2017, date of publication April 25, 2017, date of current version June 28, 2017. *Digital Object Identifier* 10.1109/ACCESS.2017.2697839

Data Fusion and IoT for Smart Ubiquitous Environments: A Survey

FURQAN ALAM¹, RASHID MEHMOOD², (Senior Member, IEEE), IYAD KATIB¹, (Member, IEEE), NASSER N. ALBOGAMI¹, AND AIIAD ALBESHRI¹

Department of Computer Science, Faculty of Computing and Information Technology, King Abdulaziz University, Jeddah 21589, Saudi Arabia

²High Performance Computing Center, King Abdulaziz University, Jeddah 21589, Saudi Arabia

Corresponding author: Rashid Mehmood (rmehmood@kau.edu.sa)

The authors would like to acknowledge the technical and financial support from the Deanship of Scientific Research (DSR) at the King Abdulaziz University (KAU), Jeddah, Saudi Arabia, under the grant number 438/611/395. The work carried out in this paper is supported by the High Performance Computing Center at the King AbdulAziz University, Jeddah.

ABSTRACT The Internet of Things (IoT) is set to become one of the key technological developments of our times provided we are able to realize its full potential. The number of objects connected to IoT is expected to reach 50 billion by 2020 due to the massive influx of diverse objects emerging progressively. IoT, hence, is expected to be a major producer of big data. Sharing and collaboration of data and other resources would be the key for enabling sustainable ubiquitous environments, such as smart cities and societies. A timely fusion and analysis of big data, acquired from IoT and other sources, to enable highly efficient, reliable, and accurate decision making and management of ubiquitous environments would be a grand future challenge. Computational intelligence would play a key role in this challenge. A number of surveys exist on data fusion. However, these are mainly focused on specific application areas or classifications. The aim of this paper is to review literature on data fusion for IoT with a particular focus on mathematical methods (including probabilistic methods, artificial intelligence, and theory of belief) and specific IoT environments (distributed, heterogeneous, nonlinear, and object tracking environments). The opportunities and challenges for each of the mathematical methods and environments are given. Future developments, including emerging areas that would intrinsically benefit from data fusion and IoT, autonomous vehicles, deep learning for data fusion, and smart cities, are discussed.

INDEX TERMS Internet of Things, big data, data fusion, computational and artificial intelligence, high performance computing, smart cities, smart societies, ubiquitous environments.

I. INTRODUCTION

The Internet of Things (IoT) [1] is set to become one of the key technological developments of our times provided we are able to realize its full potential [2]. IoT is "a global infrastructure for the information society, enabling advanced services by interconnecting (physical and virtual) things based on existing and evolving interoperable information and communication technologies" [3]. IoT was named by the US National Intelligence Council (NIC) in a 2008 report [4] among the six key civil technologies that could potentially affect US power. IoT is an enabler of ubiquitous computing envisioned by Mark Weiser. Fig. 1 depicts application areas of IoT: smart homes, warning systems, intelligent shopping, smart gadgets, smart cities, smart roads, healthcare, fire systems, threat identification systems, tracking and surveillance.

The number of objects connected to IoT is expected to reach 50 billion by 2020 due to the massive influx of diverse

objects emerging progressively [5]. The main purpose of these increasing number and types of IoT objects is to produce useful data about our surroundings to make them smarter. This is realized by providing the environments access to the information it needs through the collection and analysis of past, present and future data. The data allows optimal decision to be made about us and our environments possibly in real-time.

IoT is expected to be a major producer of big data. This data would be produced by various vendors giving rise to data as a service. Sharing and collaboration of data and other resources would be the key for enabling sustainable ubiquitous environments such as smart cities and societies [6]. The fusion of various types and forms of data, i.e. data fusion, to enhance data quality and decision making therefore would be of prime importance in ubiquitous environments. Data fusion is defined as "the theory, techniques and tools which are



FIGURE 1. Info-graphic to show IoT landscape with respect to Data Fusion.

used for combining sensor data, or data derived from sensory data, into a common representational format" [7]. A timely fusion and analysis of big data (volume, velocity, variety, and veracity), acquired from IoT and other sources, to enable highly efficient, reliable and accurate decision making and management of ubiquitous environments would be a grand future challenge. Computational intelligence would play a key role in this challenge.

A. CONTRIBUTION OF THIS WORK

The term "Internet of Things" was introduced firstly by Kevin Ashton in 1999 [8]. However, the key research and development works in IoT has started around 2010. One of the earliest and popular surveys were contributed by Atzori *et al.* [1], in 2010, where they described the fundamental building blocks of IoT and its applications. This survey was extended in [9]. Al-Fuqaha et al. in 2015 presented a survey of IoT enabling technologies, protocols and applications [10]. In [11], multi-sensor satellite image fusion methods are reviewed. The survey [12] has classified data fusion literature into three categories: data association, decision fusion

TABLE 1. Major data fusion surveys.

Surveys	Objectives and Topics
Dong et al., 2009 [17]	Focus on image data fusion methods in remote sensing, including object identification, classifica- tion, change detection and maneuvering targets tracking.
Faouzi et al., 2011 [15]	Developments and challenges of data fusion in intelligent transportation systems (ITS).
Qin et al, 2011 [18]	A brief and introductory data fusion survey for IoT.
Castanedo, 2013 [13]	Review based on a classification of the data fusion literature into three categories: data association, state estimation, and decision fusion.
Khaleghi et al, 2013 [14]	Classification of the data fusion literature based on data properties including imperfection, correlation, and insistencies
Gite et al, 2015 [19]	Focus on context awareness for multisensory data fusion IoT.
Wang et al, 2016 [20]	Classification of data fusion literature based on middleware, configuration, data processing, sen- sors, and portability.
Pires et al, 2016 [16]	Developments in data fusion for embedded sensors in mobile devices.

and state estimation. In same year, 2013, another survey paper [13] classified the data fusion literature based on sensed data properties such as data imperfections, correlation and inconsistencies. In [14], progress and various challenges for data fusion in intelligent transportation systems are discussed. A recent survey [15] has presented a critical review of data fusion developments based on embedded sensors in mobile devices with particular focus on human activity recognition. Table 1 elaborates further the surveys relevant to our paper. A summary of the main focus of each relevant survey paper is given.

It is clear that, though a number of surveys exist on data fusion, these are mainly focused on specific applications (e.g., ITS, embedded sensors in mobile devices) areas or classifications (e.g., data properties, middleware). The aim of this paper is to:

- $\checkmark\,$ review literature on data fusion for IoT with a particular focus on
 - ✓ mathematical methods (including probabilistic methods, artificial intelligence, and theory of belief), and
 - ✓ specific IoT environments (distributed, heterogeneous, nonlinear and object tracking environments).
- ✓ to explore the opportunities and challenges for each of the reviewed mathematical methods and IoT environments
- ✓ to review the emerging areas that would intrinsically benefit from data fusion and IoT (including smart cities and autonomous vehicles).

B. PAPER STRUCTURE

The paper is divided into six sections. Section II discusses the opportunities and challenges of data fusion in general as well as specific to IoT. Section III reviews the data fusion literature

based on the mathematical methods used for data fusion. These include probabilistic, artificial intelligence and theory of belief methods. Section IV discusses the literature based on the specific IoT environments including distributed, heterogeneous, and nonlinear and object tracking. Each mathematical method and environment of data fusion for IoT discussed in Sections III and IV have been elaborated with the relevant opportunities and challenges. Section V discusses the emerging areas that would intrinsically benefit from data fusion and IoT; these include autonomous vehicles, deep learning for data fusion and smart cities. Finally conclusions are drawn in Section VI.

II. DATA FUSION OPPORTUNITIES AND CHALLENGES

An enormous amount of data is produced in a quick span of time in the IoT environment. How to make this large volume of data precise and highly accurate is an open problem which needs to be solved because the quality of information plays an important role in decision making. Reliable and accurate information is critical. This can be achieved by data fusion or information fusion (terms which can be used interchangeably). Data fusion is an effective way for the optimum utilization of large volumes of data from multiple sources [11]. Multi-sensor data fusion seeks to combine information from multiple sensors and sources to achieve inferences that are not feasible from a single sensor or source [19]. The fusion of information from sensors with different physical characteristics enhances the understanding of our surroundings and provides the basis for planning, decision-making, and the control of autonomous and intelligent machines.

A. DATA FUSION OPPORTUNITIES

Data Fusion in the Internet of Things (IoT) paradigm can play a major role in its success due to the following reasons [7], [17]:

- Data Fusion makes information more intelligent, decisive, sensible and precise which is coming from multiple sensors and sources. The information from each sensor per se may not make much sense.
- A statistical benefit of fusion is obtained by computing the N independent observations; one can anticipate that the data are amalgamated in an optimal manner.
- In IoT, a big challenge is making very low power sensors which do not need battery replacements over their lifetimes; this popularizes the demand for energy efficient sensors. It has been an established fact that the sensors with high accuracy can result in the consumption of a high amount of power. To handle this issue, a set of very low power consumption sensors can be used with low accuracy. By using data fusion, highly accurate information will be created [20].
- Data Fusion can be helpful in handling the big data issues of IoT because we are fusing data from many sensors into more precise and accurate information.
- Another critical advantage of Data Fusion is that it helps to hide the critical information or the semantics which

are responsible for the fused results. Examples of this are in military applications, some critical medical areas and in intelligence buildings.

B. DATA FUSION CHALLENGES

Data Fusion has multiple challenges ahead, which are explored in various literature. Some of them are listed below:

- Data Imperfection: Sensor data is imprecise at times; it can be inaccurate and uncertain. This behavior is not infamous in wireless sensor networks. The imperfection must be dealt with effectively with the use of data fusion algorithms.
- Ambiguities and Inconsistencies: Impreciseness is not the only factor responsible for data inconsistencies; the environment in which a sensor is operating is largely responsible as well [21]. Outlier detection, replacement and data imputation are vital in IoT environment.
- Conflicting Nature: The conflicting nature of data can give rise to counter-intuitive results. The problem of conflicting data is visible more in evidential belief reasoning and Dempster's rule of combination. The data fusion algorithm must take critical care while treating conflicting data [22].
- Data Correlation and Alignment: This problem is more common in wireless sensor networks (WSNs) and can result in over or under confidence in a data fusion algorithm. An alignment problem which is also known as a sensor registration problem occurs when sensor data is transformed from every sensor's local frame to a common frame prior to fusion.
- Trivial Features: In IoT environment, applications may consist of several hundreds and thousands of sensors sensing different parameters. These sensed values in large setups such as smart cities and industrial plants consist of trivial and nontrivial data. Processing of trivial data may affect the data fusion accuracy. Thus most relevant features need to be selected before data fusion.
- Dynamically Iterative Process: Data fusion is not a static process in nature; however, dynamically iterative needs regular refinement of the estimates in a fusion environment. No Magical Algorithm: With time researches in data fusion area has advances and high performance algorithms are there now. However, it is still difficult to say that a perfect data fusion algorithm exists.

In IoT, environment information fusion can be used in various areas to enhance the IoT ubiquitous aspect. These areas are environmental monitoring, healthcare, crisis management, monitoring, controlling, tracking, intelligence gathering, and many more. Data fusion in IoT can take place at four stages: decision level, feature level, pixel level and signal level. With respect to Fig. 2, IoT data fusion can also be seen with two different perspectives. First, it can be viewed as a single hop where every sensor transmits data to the data fusion center directly. Second, it can occur by a multi-hop process, where data passes across adjacent sensors.



FIGURE 2. Diagram of Single hop and multi-hop sensor data fusion.

The multi-hop mechanism has several advantages which are discussed in Section IV.A of the paper.

III. MATHEMATICAL METHODS FOR DATA FUSION

Data fusion techniques can be classified based on the mathematical methods into three broader categories:

- Probability-based methods including Bayesian analysis, statistics, and recursive operators.
- Artificial Intelligence (AI) based techniques including classical machine learning, fuzzy Logic, Artificial neural networks (ANN) and genetic evaluation.
- · Theory of Evidence based Data Fusion methods

In Data fusion systems, there are various architectural models; however, applications constantly use their own data fusion architectures. Some of the basic and fundamental models include the Joint Directors of Laboratories (JDL) Model, Modified Waterfall Fusion Model, Boyd Model and Dasarathy's functional model [23]–[25]. These models further divide into various layers, and then the layers into sublayers. However, the core of data fusion does not lies in its architecture; it indisputably lies in the data fusion methods on which ultimate fusion processing takes place.

In this sections, we reviewed and classified the literatures based on different categories of data fusion approaches. This section is divided into three sub sections in which we reviewed literatures based on probabilistic data fusion methods, artificial intelligence-based data fusion methods and evidence theory-based data fusion methods.

A. PROBABILISTIC DATA FUSION ALGORITHMS

In this section, we reviewed some of the most recent works related to probabilistic data fusion techniques and algorithms. Probabilistic techniques are the most classical, are less complex and the most widely used for data fusion, though the accuracy might be low compared to integral techniques [19], [24]. Most of the conventional algorithms for data fusion have probability in their core. However, there are some challenges that occur for probabilistic data fusion, including: researches indicate that probabilistic data fusion systems can not represent complete information that is required for defining and depicting sensing and data fusion operations, a high level of complexities in handling nonmonotonic logic through probability and it is difficult to handle graded membership if compared to fuzzy logic in a set [19]. Bayesian theory, Markov Chain and the Monte Carlo method are some of the most studied and widely used methods in data fusion in recent times.

Probability-based data fusion algorithms are widely used in target tracking problems. The most classical example of single target tracking is the Probabilistic Data Association (PDA) algorithm. Several improved versions of PDA have been given in literature, and are very efficient in single target tracking. We also observed that the situation and complexity changes in multi-target tracking (MTT). Track validation is difficult because tracks compete with each other, therefore a more efficient Joint Probabilistic Data Association (JPDA) algorithm is used for MTT. It uses a measurement-to-track association probabilities evaluation. JPDA algorithm is a suboptimal single-scan approximation to the optimal Bayesian filter where associations are made sequentially between the tracks which are known and the latest observations [19], [24], [26]. An optimize JPDA method is proposed in [27] which is computationally tractable than conventional JPDA algorithm in applications with higher clutter density. With optimized JPDA method and simple framework in [27], performed better than some well-known MTT algorithms. Both PDA and JPDA suffer from bias phenomenon. It is observed that PDA algorithm has more biases then JPDA algorithm. In cutter environment PDA algorithm has bias phenomenon and JPDA algorithm has rejection bias and coalescence [28].

Further, a more efficient algorithm than JPDA is known as the Multiple-Hypothesis Tracking (MHT) algorithm is developed. It is a recursive algorithm. In recent years, greater attention has been paid to MHT in practical applications. MHT is important in multiple dim target surveillances, as it allows for deferred decisions. Therefore, an easy and natural accumulation of target information has taken place. Easy implementation of new tracks is initiated with simple logic [29]. In MHT, multi-target tracking is carried out by doing a likelihood evaluation which enhances its performance over JPDA [19], [24]. However, memory requirements of MHT grow exponentially as frames increase for the purpose of resolving associations [30].

The eight steps of the classical MHT tracking algorithm [29] are illustrated in Fig. 3, where, at k_{th} cycle, MHT complexity can be expressed as:

$$C_{MHT}(k) = \sum_{i=1}^{8} C_i(k)$$
 (1)

Where the computational complexity and each step complexity are denoted by C_{MHT} and C_i . In Fig. 4,



FIGURE 3. Eight steps of Multiple-Hypothesis Tracking (MHT) [29].



FIGURE 4. Comparative illustration for Probabilistic Data Association (PDA), Joint Probabilistic Data Association (JPDA) and Multiple-Hypothesis Tracking (MHT) Algorithm.

we visualize the comparison of PDA, JPDA and MHT algorithms based on their main objectives.

The Bayesian data fusion approach is the most classical fusion approach, as it is accepted and used widely for data fusion and it is the part of the core of various data fusion methods [31]. It combines multimodal information based on the probability theory [32]. The Bayesian approach constitutes of priors' definition, its specifications and posteriors computations. Several Bayesian approach-based data fusion methods are proposed in [33] and [34]. In [33], for distributed target detection, two methods have been proposed. The first one is the distributed Bayesian approach and the second method is the Generalized Likelihood Ratio Test (GLRT) for WSNs. Every sensor approximates the objects with a single bit prior to transmission to the fusion center due to energy and bandwidth constraints. The Bayesian approach proved to have better receiver operating characteristic (ROC) performance when compared to GLRT-based algorithms. The paper [33] also deals with the problem that at times the sensor number can be random and locations are unknown to the fusion center. For this problem, a fusion rule has been proposed based on the scheme that is utilized by the Scan Statistic. Whereas in [34], a method is proposed where the authors use two Bayesian trackers with various formulations. The solution handles challenges like shadows, occlusions, illumination changes, clutter, and motion changes. Thus, output is robust and with high accuracy due to the fact that the Bayesian approach is used here, incrementally calculating the probability based on new observations and uses likelihood's prior knowledge for inference.

Most of the methods available to acquire the global estimates depend on data fusion. None of these methods are perfect. Tracking algorithms accuracy is affected by common process noise. Few adaptive techniques to handle common process noise in [35]–[37] literatures are used to subdue the filtering dilemma. This is achieved by using a set of two parallel groups of filters, namely, wide bandwidth and common process noises of narrow bandwidth. However, these methods are dependent on specific target scenarios and single maneuvering targets.

For handling the target dynamics variations in fast changing scenarios and diverse accuracy requirements, an adaptive data fusion estimation scheme is required, and a hybrid algorithmic solution is needed. In paper [38], a narrow bandwidth Information Matrix Filter (IMF) is selected at quiescent levels and for maneuvering phase target tracking, wide bandwidth IMF is chosen to overcome common process noise challenges. IMF prediction depends on the propagation coefficient, which is observation independent. It is also simpler to decouple and decentralize. The paper [39] also uses the IMF technique for a novel switching algorithm which is evolved from the Bayesian classification theory [38]. It initiates in a dual-band IMF for creating a Switching Adaptive Filter (SAF) for target dynamics unpredictability. The algorithm is designed by the fading memory formulation and sum of the normalized innovations squared. This is achieved by assumption of two classification classes on statistically independent data sets. The work gives better performance results than a single IMF for the tracking of maneuvering or non-maneuvering objects.

Markov Chain, Monte Carlo (MCMC) and some hybrid methods are also used in several literatures to track objects based on the probability theory. In one of these types of work [40], a technique is used for obtaining useful information for detecting and tracking humans with lasers and RGB-D in mobile robotics. An efficient tracking algorithm based on the MCMC method was proposed for the challenging environment which includes frequent occlusions and cluttered scenes. The proposed data fusion system is competent to discover and track several humans in classical service robot scenarios.

In another similar type of work [41], a hybrid tracking algorithm is proposed which uses the Bayesian filter and MCMC sampling to manage object interactions. In this work, long-term image information is used to handle missing and uncertain face identification. The algorithm exploits both static and dynamic observations for failure assessments. The performance of the algorithm is better than other state-of-art algorithms which are not using long term observations and the Hidden Markov Model, respectively [41]. The hybrid data fusion approaches exploit the benefits of every constituent's methods. Hybrid fusion algorithms are more computationally complex than non-hybrid approaches [42]. None of these algorithms have any clear advantage over the other. However, they are more efficient and can handle more challenges with respect to non-hybrid approaches [42].

IoT middleware is an interface that integrates and facilitates the interaction between the various elements called 'Things' and internet. A very critical part of IoT middleware is event processing. The predictive fusion analytics has been studied in [43], using Bayesian Model Averaging (BAM). For proactive complex event processing, prediction analytics can play a supporting role achieved by the fusion of data. The method is used for large scale IoT applications. The method uses the expectation-maximization (EM) inference and Gaussian mixture models for basic Bayesian predictions. MCMC approximation is used for Bayesian averaging and the developed model is based on event context clustering. Results showed that the method has a high accuracy when compared to other traditional approaches. However in [43], BAM and EM still needed to be parallelized in the developed technique. For event processing in IoT, there are various challenges, such as heterogeneousness of data and devices, high volume of data, unknown event occurrence time and distributed event processing.

One of the major difficulties in sensor data fusion is that sensors regularly provide false observations which are hard to predict. Due to this fact, the data fusion systems with these spurious values from sensors must be rectified by identification and elimination. Otherwise, they will result in a high rate of inaccuracy, which would affect the final estimation. To address the spurious data problem, paper [44] used a modified version of the Bayesian approach, which automatically determines the sensor measurement and inconsistencies. In the work, three Bayesian techniques are analyzed. The first is the simple Bayesian technique, the second is the centralized Bayesian technique, and the third is sequentially fusion with a modified Bayesian technique. In the third approach, the identification of spurious data and later eliminating this sort of data, giving the highest level of accurate results among the three. In addition, the second one performed better than the first, as it has a built-in mechanism for decreasing the weighting of spurious data.

1) OPPORTUNITIES AND CHALLENGES

After reviewing the various literatures in Section III.A, we concluded that the probability based approaches for data fusion are simple, less complex and widely accepted and considered as the most classical data fusion approaches [19], [24]. The Bayesian technique can exploit prior knowledge to a great extent, thus one of the biggest advantages of the Bayesian technique-based algorithms. There are few disadvantages of the probability approach:

- Limited to the subjectivity of prior and difficult to find prior values, particularly in case of BPT.
- It is hard to represent complete information for defining and depicting sensing.
- Can result in low accuracy.

- Probability based data fusion methods complexities increase with non-monotonic logic.
- Probability based data fusion methods are not a good choice for handling graded membership.
- Not able to handle uncertainly.
- Output in prone to common process noise.
- In PDA and JPDA, computational time rises exponentially as number of targets increase specially in tracking problems.

However, with time, several improved versions of JPDA [45]–[48], MHT [49]–[51], Bayesian [52], [53], MCMC and GLRT algorithms [54], [55] are proposed to overcome the above mentioned drawbacks. Still, there is a huge window for improvement in these algorithms to address these problems, which will open the gates for future researches.



FIGURE 5. Illustration of how Artificial Intelligence enhances deci-sion making power of Internet of Things (IoT).

B. ARTIFICIAL INTELLIGENCE BASED DATA FUSION ALGORITHMS

We have to evolve IoT to be the Internet of Intelligent "Things" to make it truly ubiquitous. Sensor networks and actuators are becoming the backbone of IoT based applications such as healthcare, smart societies, military, seismic activities prediction and warning systems, intelligent transportation and tracking systems. Artificial Intelligence (AI) enables the actuators to take highly accurate and informed decisions based on the sensed data as mentioned in Fig. 5. This means AI can play a crucial role in the IoT paradigm, especially in the areas where decision making and prediction are of vital importance. Developing intelligence is a gradual process, acquired by machine learning, AI, Fuzzy Logic, Deep learning and data fusion as mentioned in Fig. 5. The process of developing intelligence for IoT-based smart societies are studied in various works [2], [6], [15],

The significance of AI in IoT can be seen in the latest happenings in the computing world. Former Apple's Siri director Luc Julia, currently the VP of Samsung Open Innovations, presented the Samsung Architecture for Multimodal Interactions (SAMI), which is part of Samsung's AI for its IoT based strategy [56]. Recently, Google acquired, Deep-Mind, an AI company, in its quest for IoT. Other recent big acquisitions consist of Boston Dynamics, a robotics company, and Nest Labs. All of this can be seen as the serious efforts of Google to boost their IoT development [57]. Similarly the Google of China, Baidu announces several IoT projects specifically focusing on AI [58]. AI in IoT is now widely used for sensor fusion, event processing and localization. In proceeding sections we will concentrate on AI based data fusion research and development endeavors.

1) SUPERVISED MACHINE LEARNING

Supervised machine learning (SML) is the approach where the algorithms learn from a set of rules known as training data. Number of independent variables represented by a response variable or dependent variable, which are often called as label. After training the SML algorithm which can also be called as classifier, prediction of response variables is carried out from given set of independent variables [59]. Various techniques of connecting intelligent things and acquiring key insights from sensed data from IoT are discussed in [60]–[63]. These studies also explains mechanisms to create intelligence environments to exploit the advantages of machine learning in IoT.

The practical application of data fusion based on Bayesian approach is pretty limited in the scenario where huge amount of data is involved. This is due to the fact that stored information is in form of quite sizable set of samples, so this data fusion approach is impractical for real-time usage where fusion centers have limited bandwidths. In [64], SVM based data fusion method is proposed to overcome the negative impact of huge datasets on Bayesian approach. Statistical learning theory used by SVM gives support for information compression with the help of representations based on optimal kernel.

Remote sensing will be one of the major application areas in near future which takes leverage from IoT infrastructure and its sensing capabilities. Better understanding of the site is achieved by fusing information from multiple sensors. In fusion research and development endeavors [65], SVM based data fusion algorithm is proposed for fusing multispectral and panchromatic data gather for the purpose of remote sensing of Shaoxing City, China. A generalize mathematical formulation of [65] is given below.

The training of SVM is given as:

$$(x_1, y_1), ..., (x_{\lambda}, y_{\lambda}), x \in \mathbb{R}^n, \{y \in +1, -1\}$$
 (2)

Hyperplane is given as:

$$(w.x) - b = 0 \tag{3}$$

Classification can be represented as:

$$(w * x) - b \ge 1, \quad if y_i = +1$$
 (4)

$$(w * x) - b \le -1, \quad if \ y_i = -1$$
 (5)



FIGURE 6. Illustration Hybrid data fusion method proposed in [66].

Where sample number is denoted by λ , input data dimensions are denoted by *n*, "." is the dot product and *w* is the normal direction of the hyperplane is the Euclidean function. Further the output shows the classification accuracy of 76.5 percent for [65]. Similarly a more recent work [66], which fuses hyperspectral and LIDAR data by using hybrid data fusion approach. Combination of SVM and Naive Bayes classifiers are used to perform data fusion in two steps as illustrated in Fig. 6. When hyperspectral and LIDAR data processed separately on the classifiers, produce 88% and 58% accuracy. However, data fusion results in 91% classification accuracy in [66]. Another feature oriented data fusion method for hyperspectral and LIDAR data is proposed in [67] which uses Random Forest machine learning algorithm.

As IoT shows promise to make our everyday life safer and prosperous by deploying safety critical applications such as smart healthcare systems, Tsunami and flood warning systems and intelligent vehicle management systems. These cyber-physical IoT systems make us aware of unwanted and dangerous situations that can arise due to the triggering of a particular event. For example brake fluid leakage in a car, this event can trigger a dangerous after-event of failure of breaks which can result in injuries to driver and co-passengers and even death. Fault detection plays a crucial role in this aspect which is critically analyzed in [68]. One of such work which addresses fault detection problem for motor using four step multi-sensor data fusion using SVM is introduced in [69] which is further explained in Fig. 7. Several machine learning based multi-sensor data fusion methods are evaluated in [70], such as K-nearest neighbor (KNN), Linear and Quadratic discriminant analysis. The data fusion is performed at feature level. Further these data fusion methods are tested on traffic management problem on multi-sensor data fusion architecture.

2) ARTIFICIAL NEURAL NETWORKS

Artificial neural networks (ANNs) have an extraordinary ability to derive meaning from complex and imprecise data. They can extract patterns and find new trends in highly complex data sets. They support adaptive learning with selforganization in a real time environment and can achieve a high degree of fault tolerance. In broader terms neural networks are basically for data fusion, as they are making data more accurate and precise by complex training and learning.



FIGURE 7. Four step of SVM based data fusion [69].

Fusing historical data by training and testing, ANNs are really helpful in predicting highly accurate values [71].

Recent travel time prediction algorithms generally require a large volume of data for the identification of algorithmic parameters. For this reason, these algorithms are less cost effective and too time consuming. To address this problem, ANNs based algorithm to compute travel time is proposed in [72]. It is a dynamic technique to predict speed, and it implements data fusion to combine speed sensors data in an expressway link. Another set of two algorithms are derived further which are Speed Integral Travel Time Calculation Method (SITCM) and Space Discretization Travel Time Calculation (SDTCM) algorithms. These two algorithms are more practical in prediction of travel time. The average prediction error is less than 10%; therefore, it can meet the requirements of on field use. Furthermore, these algorithms are simple and easy to implement. However, SITCM runs more smoothly than SDTCM, whereas SITCM is less accurate than SDTCM.

An important area where data fusion by ANNs is used effectively is wind speed forecasting. One of the such work uses ANNs to forecast wind speed by a prediction enabled by fusing the historical wind speeds together in [73]. Due to the highly complex nature of wind speed data, it is a hard task to predict future values. Accuracy in the estimation of wind power output is critical, as wind power generation is proportional to the cube of wind speed. This study provides a solution which predicts various trends of future wind speeds. It does this by proposing data fusion algorithm using several neural networks. ANNs are trained and tested by the wind data sets. The algorithm gives minimum mean absolute errors which are significant. For the prediction of monthly patterns of wind speed, two layers feed-forward back propagation (FFBP) networks with 6 in middle layer and 30 neurons in the output layers is used. For hourly prediction of wind speed, FFBP network with 30 and 12 neurons is used in [73]. Another such study [74], uses neural networks for wind speed prediction; however, the main difference between [73] and [74] is that in [74], two layered neural networks were used. However, in [73] multiple neural networks are used, which means increased complexity. Some short term wind power predictions methods are also proposed in [75] and [76].

Further ANNs learning through fusion of training data is also used in other areas like localization. The accuracy of indoor localization and navigation services based on radio signal strength indicators (RSSI) and radio frequency is on the lower side. This is due to the variable of the RSSI values. In paper [77], the multi-neural networks approach is used to solve the issue of RSSI. Bluetooth indoor localization with multiple neural networks is achieved by training and using neural networks based on user orientation. A highly accurate and cost efficient Bluetooth architecture for indoor navigation is possible.

3) FUZZY LOGIC

Lotfi Zahed in 1965 first introduced the term "Fuzzy" as he proposed fuzzy set theory and later he became famous for his fuzzy mathematics. Fuzzy logic has been used to manage concept of partial truth which range between completely true and false. It has numerous real life application as discussed in [78], such as AI, controlling, environment monitoring, gaming, electronic appliances automation, medical diagnosis and event detection etc. A technique of event detection using the fuzzy logic branch of AI in cluster Wireless Sensor Networks (WSN) for data fusion has been proposed in [79]. One of the most important applications of WSNs is environment monitoring which can be seen in larger perspective of IoT based smart societies. The system in [79] consists of multiple sensors for humidity, carbon dioxide and temperature sensing. A system based on fuzzy rules carried out the fusion of signals. Signals are collected at the cluster heads and then the fusion process takes place. The advantage of multisensor fusion is that it increases the accuracy and reliability. A important requirements of modern day societies are smart vehicle parking systems that can take leverage from IoT infrastructure to provided headache free parking management as in [80] and [81]. A fuzzy logic based data fusion algorithm is proposed in [82] for monitoring the parking space. For this purpose magnetic sensors are used and occupancy probability is computed of the corresponding parking space. The data fusion algorithm in [82] gathers information with high accuracy of the monitoring targets which results in correct decision making and has anti-interference ability. Focusing on making our societies more safer by using IoT infrastructure, in [83] an intelligent fire detection and controlling approach has been proposed which uses fuzzy based data fusion.

The use of fuzzy logic is increasing in popularity for tracking systems due to the numerous benefits mentioned above. Especially for tracking problems in the PDA and JPDA, as the number of objects increase in parallel and there is an exponential rise in the computational time. However, common sense is used in fuzzy logic to overcome this [84]. There are several fuzzy-based methods for fusion, such as the fuzzy inference correlation algorithm [85], the fuzzy double threshold track association algorithm and the fuzzy clustering means (FCMA) algorithm [86], [87].

In [87], a data fusion solution for track-to-track problems in multi-sensor and multi-target with multiple attributes is proposed. A fuzzy clustering mean algorithm (FCMA) is stated to minimize the number of tracks and by using the degree of membership of each target for determining associate duplicate tracks. For association and fusion purposes, sensor data with sensor resolution is used; the same is done to identify the most accurate sensors in the system. Monte Carlo simulation is used to show that the new scheme minimizes the computational complexity and increases the performance with respect to Euclidean clustering and Bayesian minimum mean square error method. The paper [88] concentrates on multi-sensor and multi-target distributed tracking systems, such as the ones discussed in [87].

The FCMA is robust to noise; however, selection of suitable parameters is difficult and is a challenge for providing optimal performance [87]. The work in [88] is based on track association. Association is very critical, as it is uncertain whether the particular tracks from distinct sensors represent the same target. Every sensor node performs noise measurements for distinct positions of the targets. After on board computation, data is carried to the data fusion center. This double-threshold fuzzy-based track association algorithm uses an adaptive threshold and provides less association errors, resulting in better performance.

A hybrid class of fuzzy and Kalman filter schemes are also used for data fusion purposes efficiently. One such study [89] proposes fuzzy logic-based adaptive Kalman filter (FLAKF) for multi-sensor data fusion. The measurement for the noise covariance matrix is adaptively adjusted for every local FLAKF to fit noise profile statistics for in-coming data. A fuzzy inference system is used for adaptation. The use of fuzzy logic helps to handle imprecise data. Whereas the Kalman filters tune the covariance matrix to obtain more accurate estimations. Thus, the obtained hybrid technique is more accurate. Similarly, the integration of fuzzy logic and Kalman filter is developed for data fusion in [90] for autonomous vehicles guidance.

With respect to all the literatures surveyed in Section III.B, it can be said that AI-based data fusion approaches are gaining popularity in practical use. These techniques are more accurate than other methods of data fusion and can be applied for a wide range of fusion problems. Fuzzy logic can handle sensor data uncertainty efficiently. ANNs are ideal for nonlinear systems and complex pattern discoveries are achievable. However, neural networks failed to explain how they learn to fuse from the input sensor values and AI approaches are computationally complex. In addition, some challenges remain, as a 100% correct prediction is still not achievable through fusion.

4) OPPORTUNITIES AND CHALLENGES

In recent times, the use of AI based data fusion methods is increasing. Supervised learning methods such as SVM can handles high dimensional data. ANNs and Fuzzy logic allow imprecise and contradictory inputs and handle arbitrary complexities efficiently. Fuzzy logic data fusion systems are better for handling imprecise sensor inputs and uncertainty as compared to probabilistic approaches. The ANNs data fusion approach is one of the best for nonlinear data fusion systems, and complex pattern discoveries are possible. AI fusion systems are highly accurate compared to probabilistic systems. However, they are complex and computationally expensive.

C. THEORY OF EVIDENCE BASED DATA FUSION ALGORITHMS

Dempster's, first proposed the theory of belief in understanding. Later, Shafer mathematically formalized it based on evidence-based reasoning [91]. To handle imprecision and uncertainty, the belief functions theory is a popular choice to consider. The Dempster–Shafer theory (DST) is viewed as a generalization of the subjective Bayesian probability theory (BPT). There are few differences in the DST and BPT. Firstly, BPT does not have an unknown state, whereas in DST, the unknown state could be the state of knowledge for us. Secondly, BPT assigns priors, however DST uses masses for the meaningful assignments for all the states. Thirdly, less computation is required in BPT than in DST [24].

Nowadays, the DST based [92] fusion approach is widely in use. DST is a very efficient method for feature extraction in a multi-sensor environment [92]. The ability of DST to handle uncertain and incomplete data makes it viable in multisensor systems [93]. DST based fusion systems are globally applicable and are independent of satellite image due to the empirical and reasoning parameters determination [94]. DST field is widely studied and extensive research has been done. Light Detection and Ranging (LIDAR) is a remote sensing technique that helps to determine the distance by illuminating a target with a laser and examining the reflected light. LIDAR is an important area where DST fusion can be used significantly. However minimum attention is given to the following areas: derivation of slope from LIDAR data, quantitative assessment of the classes and the interpretation of plausibility, maximum probability, support, conflict maps and uncertainty. To address the above mentioned issues, paper [95] proposed a data-driven DST application for fusing multi-sensor data to achieve feature extraction of land cover. The technique is both cost and time efficient.

A data fusion technique based on DST for fusion of digital map road signs and video detected road signs is proposed in [96]. Another similar type of work [97] which uses a fusion system for detecting traffic sign existence by merging map data and video image, whereas [98] uses DST and situation context for assigning priority for using digital maps or camera-based system. However, none of these works address false positive detection of respective camera systems. Moreover, bad weather and light conditions, and road speed limits can reduce data fusion system performance, and are rarely taken into account. Also, no quantitative data is given with respect to the computational speed of these methods. To manage these issues related to reliabilities, in paper [99], five fusion algorithms are proposed to deal with the unaddressed fusion issues of [96]-[98]. Four of them are based on information priority and the fifth algorithm is based on



FIGURE 8. Bayesian Fusion and Dempster-Shafer (DS) Fusion.

classical DST. All five algorithms were more efficient and produced more accurate outputs.

1) OPPORTUNITIES AND CHALLENGES

We concluded the above analysis of the DST literatures on two main grounds: benefits and open challenges for DST. DST obviates arduous problems in specifications of priors and with respect to uncertainty; it can be ignored on a caseby-case basis. DST is a simple approach which describes evidence patches with diverse abstraction levels and further unites the evidence. Complexity in computation is a problem in DST. Further, it cannot be considered as a proven decision theory like the universal acceptance of BPT. Primarily, BPT consists of two states which are either an event or nonevent; however, an unknown state is missing. The DST incorporates the third state called the unknown state. The DST mass functions definition is the most critical and difficult task in its implementation. The DST has been extensively studied in computer science and AI, but has never been accepted completely by statisticians [100]. In Fig. 8, we couple [42] and [101] to form a visualization of BPT and DST, which gives a succinct explanation of fusion.

IV. DATA FUSION METHODS FOR IoT-SPECIFIC ENVIRONMENTS

There are several challenges associated with data sensing. A few of the universal challenges are:

• Distributed environment of WSNs which are the subset of IoT.

- The highly heterogeneous nature of IoT due to various heterogeneous devices and data.
- Nonlinearity and tracking issues like: multi-target tracking (MTT), cost effectiveness, error mitigation, asynchronous and track-to-track (T2T) problems.

In this section, we review and classify literature based on the above mentioned challenges faced by data fusion methods. This section is divided into four parts: distributed environment challenges, heterogeneous environment challenges, nonlinear environment challenges and tracking environment challenges.

A. DISTRIBUTED ENVIRONMENT

Theoretically, distributed data fusion is a fairly developed concept in the last few decades, and has been studied in depth [102]. However, the practical implementation of these techniques is still a challenging task in the IoT environment. The distributed algorithms do not have any centralized control. They can also be referred to as sub branches of parallel algorithms. The fundamental elements of a distributed information fusion system in IoT are the sensors and processors. Sensors are responsible for data generation by observing the operating environment. Processors are responsible for fusing the data. In this section, we review and classify the distributed data fusion algorithms in WSNs which can be seen as subsets of IoT. In the distributed environment, the fundamental estimation algorithm is the distributed Kalman Filter (KF) algorithm and is highly scalable. In various practical problems, KF based algorithms play significant roles. KF can be considered a prediction corrector filtering algorithm which is achieved by evolution or state propagation and data updating. We can consider KF as a Bayesian fusion algorithm, as proved in [101]. The various areas where distributed KF have been implemented include weather and environmental monitoring, surveillance, tracking and medical areas.

Most of the algorithms in this class are static in nature; however, the real environment network topology is dynamic. This change in network topology may be caused by the failure of the node, which is a result of reduced energy. The study [103] deals with such types of networks. The work based on simulation of 200 sensors is distributed randomly in a dynamic environment. The Central Kalman filter (CKF) is used for evaluation of Distributed (KF) in the work. The proposed algorithm behaved very close to CKF. The Distributed (KF) algorithm-based sensor can run up to six neighboring sensors. KF implementation in a distributed manner holds the advantage of gaining matrices computational cost when compared to CKF. The other advantage for Distributed (KF) is that the scalability is easier, which is required in a dynamic environment like IoT.

In IoT, energy efficiency plays a critical role, since we have hundreds of sensors operating together. The system must be energy efficient. Otherwise, a lot of cost is incurred on energy consumption by the sensors. A Cuckoo Based Particle Approach (CBPA) is used by the random deployment of

nodes based on static clusters with the Cuckoo Search in a distributed WSN. After the cluster heads are selected, the data is collected, aggregated and forwarded. A generalized particle approach algorithm is used and the data is forwarded to the base station. The network energy consumption problem transforms into dynamics and kinematics, completed by the Generalized Particle Model Algorithm (GPMA). The outcome of this hybrid approach results in a reduced low cost energy efficient technique. A high degree of consistency with a low level of complexity due to the sub-optimal algorithm is also achieved [104]. With the same goal, work [105] aims to reduce energy cost related to the distributed data fusion process. It is achieved by providing an approximate solution known as P2lace. This is done in two phases; in first phase, a task graph partition takes place, and in the second, task graph placement occurs.

Another way to reduce energy consumption in the distributed WSNs fusion systems is to put some sensor nodes in a sleep mode for some time, while the remaining sensor nodes are active. A distributed data fusion algorithm for sensor nodes has been proposed in [106]. The algorithm is time driven, which performs network data aggregation and is accomplished by nodes scheduling and batch estimation. The scheduling of the least sensor nodes is done in the clusters. This is done because of two reasons: firstly, to meet the conditions of the acquisition cycle, and secondly, to reduce the time that the sensor nodes are in a working state. In parallel, data fusion is introduced based on batch estimation. Output produced by the algorithm shows reduced network energy usage and improved reliability. This fusion system can be efficiently used in health monitoring applications.

A similar study to [106] is proposed in [107] which is based on cluster formation and reduces the energy cost of data fusion. However, the major difference in the contribution of these two works is that the data fusion technique of [107] is based on a multiple path selection with a packet delivery ratio on the higher side. A few more ways of achieving energy efficiency are proposed in [108]. Two energy planning algorithms for progressive estimation are proposed. The authors also computed energy cost for consensus estimation; it is a technique for the distributed fusion in the multi-hop sensors model based on peer-to-peer networks. The set of algorithms evolved from the following principles to achieve energy efficiency [108]:

- Multi-hop are highly energy efficient because they refrain from long distance data transmission.
- By progressive data fusion which hops through sensors resulting is low energy consumption.
- Reducing energy by the predetermination of transmission energy. It is done with the help of the prior knowledge of all channel state information and a routing tree.

Scalability is a challenging task in a distributed environment such as IoT, which is also heterogeneous and dynamic. In an environment like IoT, a significant issue is that suddenly a number of sensors can awake, adding several nodes in WSNs. The data fusion algorithm must be efficient to deal with these kinds of situations. To address this issue, [109] discusses and analyzes several sub-optimal algorithms. These algorithms include the following:

1) CHANNEL FILTER

It is a simple data fusion approach. Only the first ordered redundant data is taken into account. Every channel has a pair of agents, a transmitting agent and a receiving agent. Redundant information is removed by the transmitting agent. However, ad-hoc WSNs transmitting data sometimes does not reach the other end. Therefore, the receiving agent can do the task of transmitting agent in the dynamic ad hoc WSNs. Channel Filter fusion equation is given as:

$$p(x) = \frac{p_1(x) p_2(x)/\bar{p}(x)}{\int p_1(x) p_2(x)/\bar{p}(x) dx}$$
(6)

In Equation (6), $p_1(x)$ and $p_2(x)$ are the density function fusion probabilities and $\bar{p}(x)$ is the previous density function received. The advantage of this algorithm is that there is no need to maintain a high volume of history of past activities. Though one disadvantage is that during the filter, dependent information is removed. However, this effect can be minimized if the time between current processing and when redundancy occurred is too long.

2) Naïve FUSION

It is one of the simplest data fusion techniques. It is anticipated that the dependency between the density functions is minimal; however, the technique is unreliable. Due to the lack of past information, over-confidence can occur. The naïve fusion equation can be written as:

$$p(x) = \frac{p_1(x)p_2(x)}{\int p_1(x)p_2(x)dx}$$
(7)

3) Chernoff FUSION

In unknown dependency distribution, the Chernoff technique can be used. Theoretically, two arbitrary density functions can be combined using Chernoff fusion in a log linear fashion. However, fused density may be distracted. Another disadvantage is that extensive computation is required. Chernoff equation is given as:

$$p(x) = \frac{p_1^w(x) p_2^{1-w}(x)}{\int p_1(x) p_2(x)/\bar{p}(x) \, dx}$$
(8)

Where w ϵ [1, 0]

Though there are several studies based on data fusion algorithms in a distributed environment, there are still several challenges that remain. As stated in [109], the data fusion process is described mathematically by the set theory in the given equation:

$$\Phi(.|I_i) = \Phi\left(.|\bigcup_{i=1}^{n} I_i\right) \frac{1}{C} \prod_{i=1}^{n} S_i^{(-1)^{i-1}}$$
(9)

Where S_i is the *i* event probability combinations, the alternating division and multiplication of the joint probabilities from (6), get rid of conditional dependencies form of shared information in the datasets.

4) OPPORTUNITIES AND CHALLENGES

In the theoretical distributed data fusion, removal of duplicate information is simple. Duplicate information identification for distributed fusion systems is arduous when this theory is put into practical use. In distributed fusion, it is difficult to recognize correlated information originating from past fusion events and to get the values of previous data sets. IoT is highly heterogeneous, both with respect to devices and data, which makes distributed fusion challenging. Due to its dynamic nature, distributed WSNs change their shape and size frequently, therefore network scalability is an issue of concern in distributed fusion systems. Heterogeneity is also a challenge for IoT. In the next section, we analyze and review the literatures of heterogeneous data fusion algorithms.

B. HETEROGENEOUS ENVIRONMENT

IoT environment is not always homogeneous; therefore, heterogeneity is not rare in this diverse environment in terms of both devices and data. One of the major difficulties which heterogeneous fusion systems face is due to the different feature spaces of data sets. In heterogeneous systems, data sets are generally represented in several feature spaces. This makes it difficult to analyze relationships among different data, even when the data sets are related to each other semantically.

As the solution of the above problem, a graph embedding framework in [110] is used to deal with space alignment issues in heterogeneous fusion systems. The proposed framework converts every data set into a graph and zero distance assignments between corresponding pairs, which eventually results in a single graph. A non-metric multi-dimensional scale is proposed which uses rank order. The advantage of using rank order is that this type of fusion system can manage alignment as well as deformation. This technique proved efficient and better than the existing constrained Laplacian eigen-maps, tensor decomposition and Procrustes analysis methods.

A Bayesian filtering-based method for heterogeneous systems is addressed in [111]; the state space model is used for locating the estimation of radio measurement and speed sensors. The tracking problem is divided into several local constraints with mutual interactions with factor graphs by message passing. During every iteration, the messages are passed efficiently with reliable information. This takes place between the prediction and the correction phase. To obviate the effects of error propagation due to speed sensor variances, the algorithm used a fixed-lag smoothing technique which relied on the past and future data of a particular point. The algorithm is less complex and shows high accuracy, thus tends to minimize the computational load. Several additional schemes are suggested in [112] and [113] to manage heterogeneous bio-medical data using the Bayesian technique. Various fuzzy logic-based algorithms and hybrid algorithms-based on the combination of fuzzy logic and the Kalman filter are better and more effective for heterogeneous sensors systems, which measure the same parameters with diverse dynamics and noise statistics [84].

Fusing data from heterogeneous observations promises to find complex multivariate relationships among the data sets. In the analysis of multivariate data from the two sets of variables for the extraction of correlated features, there are two fundamental methods: canonical correlation analysis (CCA) [114], [115] and Partial Least Squares regression (PLS) [116], [117]. PLS is feasible when two sets of variables are dependent on each other or one set of variables holds the explanation of the other [114]. On the other hand, CCA is more viable when two sets of variables are symmetrically related to each other. CCA is widely studied for explanatory multidimensional statistical analysis. The main goal of CCA is to determine the linear combinations of every variable in the given data sets; this is accomplished when the correlation is at its maximum among the linear combinations.

The major disadvantage of CCA is that the data fusion output degrades dramatically with noisy datasets [114]. In paper [118], a novel method is proposed called Noise-Outliers Removal Algorithm (NORA). NORA handles noisy data sets in heterogeneous data fusion systems in which [114] (Gonzalez, 2009) lacks. NORA is used to filter features and non-informative data points before the execution of CCA. Specifically in [118], NORA is used for preprocessing neuropsychology and Magnetic Resonance Imaging (MRI) prior to CCA execution for identifying the association between them. Several other methods were proposed to handle heterogeneous data using Bayesian analysis [119], DST [120], multiple-metric learning method [121] and hybrid data fusion method based on probability and DST in [122].

1) OPPORTUNITIES AND CHALLENGES

Heterogeneity in IoT environment is a challenging issue to handle during system integration due to disparate sources of data. These data sources cannot be combined as it is, methods are required to transform heterogeneous data to homogenous space. Further heterogeneous datasets add uncertainty. Complex multivariate relationships among the datasets. However, fusing data from heterogeneous observations promises to find complex multivariate relationships among the data sets.

C. NONLINEAR ENVIRONMENT

Nonlinear time-varying sensing also brings formidable challenges to multi-sensor data fusion. Nonlinearity can result in less accurate estimations. In [123], it is shown that an improved estimation can be produced in acute nonlinear systems at the fusion center with the assistance of optimal data fusion in a multi-sensor setup. An optimized algorithm according to each sensor's channel conditions for power allocation of respective sensor nodes is proposed. It performs dynamically for the power assignment of all sensor nodes subsets. The technique is based on a semi-definite program (SOP) [124]. Therefore, it guarantees the best available state estimation by intending to reduce mean square error values.

The Extended Kalman Filter (EKF) has been one of the most comprehensive algorithms in nonlinear tracking environments [125], [126]. However, more accurate values have been obtained from unscented Kalman Filter (KF) than EKF with the help of approximation statistics [127]. In practice, EKF has three disadvantages [128]:

- Unstable filters can be produced due to EKF linearization.
- Using EKF, linearization can only be performed if the Jacobean matrix exists.
- Linearization using EKF is highly difficult to implement due to the fact that Jacobean matrices derivation is nontrivial in most of the cases.

An unscented linear fractional transformation (LFT), which is more efficient than unscented KF, is proposed in [129]. The LFT transforms the nonlinear system to a tantamount linear model and an unscented transformation handled nonlinear structure. Further [123] broaden the LFT technique to a multi-sensor environment using the Bayesian approach.

1) OPPORTUNITIES AND CHALLENGES

KF based data fusion methods are a popular choice for nonlinear environment. It is simple, less complex and easy to implement and widely accepted. However not viable with spurious observations. Exposure to outliers can result is KF breakdowns especially in sensor dense environment like IoT. It Accuracy can be questionable with respect to AI and hybrid methods and extensive computation is needed in KF if too many sensors are involved. KF are better for linear systems and conditional independence is involved. Fuzzy logicbased data fusion methods are really useful in non-linear and multi-variable systems. They are also easy to modify. A fuzzy system-based on rules fuses raw data acquired from sensors. Along with this fuzzy predictor to make data fusion highly accurate for too sensitive applications, this fusion system can work with high bandwidth and efficiency [130]. In nonlinear systems, the update problem is also critical. Regardless of several works, nonlinearity is still a complex task to manage in multi-sensor data fusion systems.

D. OBJECT TRACKING

The tracking of objects is one of the oldest areas where the use of data fusion comes into the effect. Data fusion in tracking domain plays an important role in military applications, robotics, wireless systems, and transportation. In [131], various positioning principles and the interaction between IoT and objects in tracking and positioning systems are described. Some of the possible benefits of tracking and surveillance fusion systems include correct target selection, locating the threat, identification of unidentified and unauthorized moving objects in high-level security zones and timely decision making. In this section, we review and classify various challenges in tracking data fusion algorithms, including multi-target tracking (MTT), cost effectiveness, error mitigation, asynchronous and track-to-track (T2T) problems.

A multi target tracking algorithm uses a scene adaptive hierarchical data association. This scheme adaptively determines the features with high reliability in the respective scenes for the given targets. With the help of reliable features, hierarchical feature spaces have been created, and different layers data associations take place. The algorithm works efficiently and effectively in both indoor and outdoor systems [132], whereas in [133] a MTT algorithm based on the sensor allocation problem and a set of sensors are dynamically identified. Later, track data fusion objects are tracked and collaboration is performed. As target objects move from time to time via sensor assignments, the formulation of the problem in regards to constrain optimization for maximizing tracking performance for respective targets has been done. The algorithm then performs an iterative sub gradient search which is near optimal for the integer programming problem. The stated solution is cost effective and scalable. The paper [134] also discusses the coverage guarantee and energy efficiency. Another multi-object tracking fusion technique uses ego vehicle odometry, image and radar where data fusion takes place at a high level which gives highly reliable results. This scheme also locates stationary objects and can perform width estimation. These algorithms are implemented using the application SASPENCE. This approach proves to be highly robust in difficult environmental conditions [135].

Data fusion in tracking and positioning applications is also popular in the automobile industry. Global Positioning System (GPS) data fusion along with data from Inertial Navigation System (INS) is popular in positioning systems. Although INS applications are highly accurate, the installation of INS is costly and time consuming. Vehicle positioning systems estimation must be highly accurate, reliable and with the information continuity provision. Low cost GPS receivers are commonly used in traditional automobile applications. These systems are not highly accurate or reliable, and do not provide a guarantee of information continuity provision during GPS error.

To mitigate the errors in GPS, several Bayesian filter-based data fusion algorithms have been discussed in past literatures; the performance of Bayesian filter estimations largely depends on the selection of a process model. The dynamic driving conditions must be taken into account by the Bayesian filters. An Interacting Multiple Model (IMM) based data fusion positioning algorithm is proposed in [136]. The IMM filter uses GPS and embedded sensors to adjust with respect to different dynamic driving conditions. The kinematic vehicle model and a dynamic vehicle model are integral parts of the IMM filter in the study. The algorithm is cost effective, accurate and reliable in dynamic driving scenarios. This is due to the fact that the IMM filter uses N parallel KF and is an approximation algorithm. From the group of various multiple model filters, the IMM filter algorithm accomplished

$$Cov [X,Y] = \begin{pmatrix} E [(X_1 - E [X_1]) (Y_2 - E [Y_2])] & E [(X_1 - E [X_1]) (Y_2 - E [Y_2])] \\ E [(X_2 - E [X_2]) (Y_1 - E [Y_1])] & E [(X_2 - E [X_2]) (Y_2 - E [Y_2])] \\ E [(X_3 - E [X_3]) (Y_1 - E [Y_1])] & E [(X_3 - E [X_3]) (Y_2 - E [Y_2])] \end{pmatrix}$$
(10)

first-order generalized pseudo-Bayesian (GPB1) estimator computational capabilities. Specially, GPB1 is a very powerful technique for behavior evaluation.

Locating critical targets is one of the most important applications of data fusion for the military and other highly sensitive security areas. High rates of detection probabilities and low error rates must be present for stringent performance. Data fusion is an effective approach for ameliorating detection performance by enabling collaboration via sensors with limited reliability. Military sensor network deployment is costly; therefore, it is preferable that optimal location placement of sensors are used to reach utmost performance. This is computationally complex, non-convex and a nonlinear optimization problem. Based on the probabilistic data fusion model, a fast and efficient sensor placement algorithm is proposed [137]. It performed better compared to other algorithms in the literature.

Tracking data fusion systems use association and estimation. The IoT system consists of multi-sensors and is also designed for multi-targeting. The two broad types of association used in these systems are measurement to track (MT) and track-to-track (T2T) associations. The major difference between the two is that sensor level implementation is carried out in MT, and implementation is done at data fusion center level in TT. The association in T2T fusion is crucial. Due to false and missed tracks, random errors and sensor bias make it more perplexed. A Bayes joint decision and estimation (JDE) is optimally implemented in parallel with sensor bias to obtain a simplified JDE. With the checking of association error sensor bias, this scheme results in improved accuracy [138].

A complex asynchronous problem arises in T2T fusion during tracking of fast moving objects; however, it is insignificant for slow moving objects. To solve this problem, authors in [133] proposed a solution; the solution is executed in three different phases. In the first phase, estimation takes place at the fusion center. Recording the actual time corresponding to the fusion center time reference after acquiring the sensor data is done in first phase. In second phase, predictions are used by the fusion center to shift the received data, which will then start the next fusion cycle. This step synchronizes the data required for real time un-correlated track-to-track fusion. In the third and last phase, pseudo-synchronized data with a linear minimum variance un-biased estimator algorithm is used to fuse all the sensors' data.

An asynchronous KF is often used in T2T fusion without regard to its known drawbacks as mentioned in [138]. A comparative analysis has been done in [139] of three well-established T2T algorithms: Cross covariance [140], Covariance intersection [19] and Covariance union [141] with asynchronous KF to access the performance for the T2T fusion problem.

The Cross covariance T2T fusion outperforms the other two with respect to root mean squared errors (RMSE) and also the run time of Cross covariance is minimum. The Covariance intersection gives the second minimum runtime, where RMSE is just over asynchronous KF. In the Covariance union, RMSE is approximately equal to Cross covariance but with a high computational cost. The reason why asynchronous KF for T2T fusion gives minimal performance is due to the following facts:

- The unified KF object list is asynchronously updated at every instance when new sensor object list arrives. This is principally incorrect for KF as sensor objects of every sensors are correlated on a temporally basis which is the outcome of previous filtering and the same correlation phenomenon is observed for tracked objects.
- Implementing KF to already KF filtered data will produce additional phase delays due to low-pass property of KF.

The mathematical formulations of these methods are given below:

a: CROSS-COVARIANCE EQUATION

The cross-covariance matrix [140], [142], between X and Y is a M X N matrix denoted by Cov [X, Y], where X and Y are the random vectors, E is expectation operator and $X = [X_1X_2X_3]^T$ and $Y = [Y_1Y_2]^T$ [see (10), as shown at the top of this page].

b: COVARIANCE INTERSECTION EQUATION

Covariance intersection algorithm [19] is used for fusing two or more state variable estimates in a KF with unknown correlation. a and b are two known information items which are to be fused into information item c. Item a and b have mean/covariance \hat{a} , A and \hat{b} , B, however the cross correlation is unknown. The mean and covariance for item c has been given by covariance intersection update as:

$$C^{-1} = \omega A^{-1} + (1+\omega) B^{-1}$$
(11)

$$\hat{c} = C(\omega A^{-1}\hat{a} + (1+\omega)B^{-1}\hat{b}$$
 (12)

Here ω must be computed for reducing the norm.

c: COVARIANCE UNION EQUATION

The covariance union method proposed in [141], allows to fuse two tracks, even if the difference of the state estimates exceeds the covariance presented by at least one track. A new state vector $\hat{\varepsilon}_C$, is used to obtained u fused estimate. Fused covariance matrix is denoted by P_c which exceeds both P_a and P_b . The fused estimate $C = \{\hat{\varepsilon}_C, P_c\}$ is



FIGURE 9. Heterogeneous multi-object tracking scenario in real time.

determined by: $U_a = P_a + (\hat{\varepsilon}_c - \hat{\varepsilon}_a) \cdot (\hat{\varepsilon}_c - \hat{\varepsilon}_a)^T$ and $U_b = P_b + (\hat{\varepsilon}_c - \hat{\varepsilon}_b) \cdot (\hat{\varepsilon}_c - \hat{\varepsilon}_c)^T$ where $P_{ab} = \max(U_a, U_b)$ and $\hat{\varepsilon}_c = \arg\min(\det(P_{ab}))$

T2T fusion faces some challenges: (1) in T2T, asynchronous data fusion in real-time tracking is a hard task to achieve, (2) the problem of dependency between the estimation of tracks and states of various sensors in data fusion center, and (3) the problem of data redundancy and common process noise which may be due to sensor biases and correlation [133], [143].

Also, tracking data fusion algorithms face challenges, such as fast moving objects, high density clutters and asynchronous systems. There are several other challenges for tracking data fusion algorithms with respect to IoT: (1) multipath mitigation in radio frequency, (2) short term and precise synchronization and pre-synchronization is required, and (3) IoT cannot render the use of localization cells for a single node exclusively because it is not feasible.

In Fig. 9, we illustrated an info-graphic to show several tracking scenarios. In [144], the deficiencies are explained of current commercial airplane tracking systems. Recently, the Malaysian airlines flight MH370 crash site could have been located if it was tracked by multiple trackers like ATC and satellites; then fused data could be used to identify the plane's location. This is just one real life example showing how useful data fusion can be.

1) OPPORTUNITIES AND CHALLENGES

With time, multiple target tracking is getting popular in terms of applications areas which are human-computer interaction, motion-based identification, automated surveillance, traffic monitoring and vehicle navigation [137]. Past works in multiple target tracking combine the set of features for uniquely identifying the objects. However, it is inclined to give high computational costs with error accumulation, especially in the IoT environment. Today's network camera setups are complex with fast changing scenarios. Therefore, fixed techniques for feature selection are not feasible. There are some major challenges for multi-object tracking data fusion algorithms in the IoT domain [132], [133], [134], [145]:

 The complexity of data association in IoT based systems is high, as critical association decisions are performed locally. This can be optimal for local domains but not globally.

- Fallacious association decisions are impossible or difficult to undo once they occur.
- The asynchronous fusion problem is inevitable in high speed target tracking, especially in T2T fusion.
- Object occlusions and complex shapes of objects.
- Information loss during 3D to 2D projection.

V. EMERGING DATA FUSION TRENDS

Recent research and development endeavors on data fusion show its evolution from conventional domains to more futuristic application domains such as infotainment systems, human activity recognition, connected and autonomous vehicles. In this section, we will focus first on recent application trends where data fusion can be applied and how deep learning can enhance data fusion.

A. AUTONOMOUS AND CONNECTED VEHICLES

An autonomous vehicle is self-driven, which has its own intelligence to understand the environment around it to drive steadily and safely. Various machine learning algorithms are used to perceive the driving environment by processing data from one type of sensor (source) such as: RGB camera, LIDAR, GPS etc. Another way to achieve driving environment perception is to combine data from multiple sensors for accomplishing a single objective. For example: combining GPS and camera images to predict safe driving distance to another vehicle on the road. Use of data fusion is quite old in autonomous vehicles. However, not much work has been done in this field prior to the year 2010. In recent years self-driving cars research and development works gained popularity and enormous interest are shown by Information technology and automobile industry titans for developing "intelligence to drive". This paved the way for developing state-of-art data fusion technologies for autonomous driving.

Fusing the multiple inputs into a single output is a complex problem but the outcome tends to be more accurate than single sensor data analytics as achieved by proceeding literature. For example in [146] authors use combination cameras and LIDAR for driving scene understanding by labeling image segments where as in [147] camera and laser are used to create object grid maps. In [148] and [149], single source data is used to identify pedestrians. Addressing the same problem, however using multiple data sources, a smoothingbased depth up-sampling technique for pedestrian detection is proposed in [150] which fuses camera and LIDAR data. Similarly in [151] authors uses knowledge of object classes to detect pedestrian, car obstacles and bicyclists. A multi-layer perceptron (MLP) classifier is used in [152] to Identifying, explaining and tracking independent moving objects. Combination of stereovision, speedometer and LIDAR data is used and feeded to MLP in [152]. Hane et, al use cameras image and wheel odometry for extracting static obstacles [153] whereas in [154] DST is implemented to fuse sensors data and identify the obstacles using camera, laser radar, GPS, perception sensor. As roadway condition

knowledge is vital for safe and smooth driving, in [155] camera and LIDAR data combined together for performing lane estimation. None of these literature mentioned above have any clear advantages and disadvantages over the other due to the fact that, driving environment conditions, learning models and data used for testing are distinct.

Looking at the pace of technological advancement in autonomous driving and successful deployment of IoT infrastructure for creating future smart societies in developed economies around the world, increase the thrust to enhance "Intelligence to drive" to newer level in near future. This can be achieved by connecting autonomous vehicles together by using IoT infrastructure and let them talk and share their data and intelligence. The recent deadly crashes of Google [156] and Tesla [157] self-driving cars put a question mark on maturity of "AI to drive" developed by these technology titans. Data fusion methods as discussed in Section V.A are fusion of data from within the sensor board of the autonomous vehicle, no external data is used other than GPS data. To the best of our knowledge, there is hardly any focus on developing data fusion methods for an autonomous vehicle so that it can fuse information which it is acquiring from other autonomous vehicles which are connect to it using IoT infrastructure.

However, to develop data fusion methods in future for autonomous connected vehicles can bring a completely different set of challenges. We believe that this will further improve the driving environment understanding for autonomous driving.

B. DEEP LEARNING FOR DATA FUSION

Deep learning (DL) branch of machine learning is getting significant attention. Gartner classified deep learning as one of the top 10 technology trends which have significant impact on the strategic planning of most organizations for the year 2016 [158]. Deep learning is a representation learning model that mimics the neural system of humans. It takes raw data as input and automatically discovers representations required to do predictions. Deep learning tries to model higher level data abstractions. Deep learning model can have several layers between input and output which helps it to think. An intriguing fact about deep Learning is that, layers of features are learned from data automatically. LeCun the director of AI research at Facebook in his famous Nature review publication on deep learning [159], stated that deep learning will see many near future successes because of two important factors: (1) it needs very little engineering by hand and (2) it inherently benefits from the increases in the amount of available computational resources and data. In another review, Wang and Raj examine deep learning evolution from its initial stage to present [160]. Deep learning reviews in [159] and [160] addressing the same topic but with different objectives. Paper [159] critically explains the present state-of-art developments in deep learning with its application area. Further giving future possibilities in deep learning whereas in [160] authors discuss the past and Present of deep learning and how it transformed from classical ANNs to deep learning. Deep architectures and mathematical formulation of deep learning models are comprehensively discussed in [161] and [162] which can be referred for further information.

Classical machine learning methods such as SVM, Random Forest etc. are widely used for fusing information from multiple sources. However, with a very slow pace deep learning branch of machine learning is making inroads into data fusion domain. DL models are successfully applied for multimodal learning in [163]–[165] which involves complex learning procedures. In [163], audio and video features are correlated to extract relating information, whereas in [164] authors proposed deep learning model for deep multimodal fusion of discrete events. Further in [165], deep fully convolutional neural networks are used to for multimodal fusion of Earth observation data. A deep belief networks (DBNs) based data fusion method is proposed in [166] for fault monitoring of ball screws which identify critical ball screw health related patterns. Another work [167], uses DBNs for data fusion. It uses 128x128 Dynamic Vision Sensor and 64-channel AER-EAR silicon cochlea data as inputs. The proposed data fusion method able identify digits even in the presence of distractions [167]. There are handfull of few more deep learning based data fusion methods proposed for different application domains in the following literatures such as data fusion for activity recognisation in [168], data fusion for network traffic [169], audio and visual data fusion in [170] and pedestrian detection in [171].

DL branch of machine learning has several issues. Firstly, they are resource consuming,. This means they need significant processing power and memory. But in recent days hardware cost is decreased and tends to decrease in future, so the resource consuming aspect of deep learning can be overlooked. However, Incase of data fusion system resides within a mobile device rather than a power system, in this scenarios applicability of deep learning is arguable. Secondly, deep learning methods are vulnerable to adversarial samples at training phase. Thirdly, deep learning methods need tons of data to get trained but a human can perceive environment just from one sample or few samples. Despite these issues deep learning is proving its worth in several application areas such speech recognisation, image analysis, autonomous driving and pattern recognisation.

C. MOBILE DEVICES, SMART CITIES AND SOCIETIES

With the daily increasing number, people are actively migrating from rural areas to urban areas in search of better life. This is eventually putting an extraordinary burden on urban infrastructure and services. This means that most essential resources are getting scared such as water, power, medical facilities and transportation etc. Even the basic services in big cities around the world are in turmoil. There is an urgent need to manage available resources very efficiently and to predict highly accurate future requirements for growing urban population. How to achieve this goal? The answer lies in the premise of Smart City which looks intriguing [172], [173]. The "notion of smart cities can be extended to smart society; i.e. a digitally-enabled, knowledge-based society, aware of and working towards social, environmental and economic sustainability" [6]. Smart City uses IoT infrastructure, where several thousands of sensors are sensing our environment. Further, the sensed data which is huge in size and in various formats aid the actuators to take the conducive actions. As already discussed in previous sections by fusing data from multiple sensors, improved accuracies and inferences can be achieved than single data source. For Smart Cities, data fusion is vital due to its property to make inferences based on heterogeneous data sources with high accuracies and further, it creates an urban knowledge base.

Data Fusion has been studied for various applications of Smart Cities such as traffic management, warning systems, event detection, healthcare, power supply management and pollution control. Data from European Project SmartSantander is fused to correlate traffic patterns in Santander City with respect to the temperature [174]. Whereas in [175], intelligent road traffic application is proposed by combining data from roadside sensors at fusion center in an energy efficient and cost effective manner. In Smart City, different sensors can record different attributes of a particular event. For example, static (fixed) sensors can record attributes like "where what and how" but failed to describe "who". Whereas wearable sensors can record the attribute "who". A data fusion algorithm is introduced in [176], for combining such data streams which sense the same event but distinct attributes of it. Several other works which are aimed at different applications of Smart Cities by implementing data fusion such as water management [177]-[179], social big data [180], smart power supply and management [181]-[183], intelligent traffic management [184]–[186], and smart healthcare [187]-[189] are introduced to take leverage from IoT infrastructure. Other works relevant to various services in smart cities or societies include emergency management systems [190], [191], [208], IoT-based proposals for improving cultural virtual reality based traffic event simulations [192], autonomic mobility systems [193], [194], urban logistics [195]–[198], location based services [199], multimedia performance analysis over networks for smart cities [200], [201], [203], crime-sourcing [204], community resilience [198], vehicular ad hoc networks that could be used for mobility and data forwarding (such as fog) purposes [191], [205], green computing for mobiles [206], cloudlets [207], artificial intelligence [208], intelligent mobility with social conscience [209], and location based services with data privacy [199], internet of cultural things and similar proposal [210]-[212], and city planning [8].

The increasing use of mobile devices like smartphone, smart watches, and tablets etc. is on a rise, the various applications on these devices need to access a lot of information and of a different type for a better understanding of our contexts. This, in turn, increases the demand for platforms that facilities sensor data fusion. This power the next generation of smart mobile devices for future smart societies. A smartphone consists of an accelerometer, gyroscope, digital compass, ambient light sensor, and a proximity sensor which holds the ability to provide better context-awareness. Pires et al, in their work comprehensively reviewed data fusion methods applied to embedded sensors in mobile devices for human activity recognition [15]. Further in [213] and [214], they proposed several fusion methods based on mobile devices to recognize various daily living activities. Very little work has been done on data fusion methods which are particularly focused on the daily use of mobile devices. We believe in future there will be far greater use of mobile devices, hence better contextaware application will be needed, where data fusion can play a major role.

Today, the total and large scale implementation of smart cities around the world are in their infancy period due to financial and technological limitations. Real challenges are ahead when in near future the above-discussed applications of smart cities are sum-up together for transforming our present day social and technological environment to vibrant smart societies.

VI. CONCLUSION

The Internet of Things (IoT) is set to become one of the key technological developments of our times provided we are able to realize its full potential. The number of objects connected to IoT is expected to reach 50 billion by 2020 due to the massive influx of diverse objects emerging progressively. IoT hence is expected to be a major producer of big data. Sharing and collaboration of data and other resources would be the key for enabling sustainable ubiquitous environments such as smart cities and societies. A timely fusion and analysis of big data, acquired from IoT and other sources, to enable highly efficient, reliable and accurate decision making and management of ubiquitous environments would be a grand future challenge. Computational intelligence would play a key role in this challenge. A number of surveys exist on data fusion. However, these are mainly focused on specific application areas or classifications.

In this paper, we aimed to review literature on data fusion for IoT with a particular focus on mathematical methods and specific IoT environments. The mathematical methods discussed included probabilistic methods, artificial intelligence, and theory of belief. The environments discussed included distributed, heterogeneous, nonlinear and object tracking environments. The opportunities and challenges for each of the mathematical methods and environments were discussed. Emerging areas that would intrinsically benefit from data fusion and IoT, autonomous vehicles, deep learning for data fusion and smart cities, were also discussed. The opportunities and challenges of data fusion in general as well as specific to IoT were provided. To the best of our knowledge, currently, no such survey exists.

REFERENCES

- L. Atzori, A. Iera, and G. Morabito, "The Internet of Things: A survey," *Comput. Netw.*, vol. 54, no. 15, p. 2787–2805, Oct. 2010.
- [2] F. Alam, R. Mehmood, I. Katib, and A. Albeshri, "Analysis of eight data mining algorithms for smarter Internet of Things (IoT)," *Procedia Comput. Sci.*, vol. 98, pp. 437–442, Dec. 2016.
- [3] Overview of the Internet of Things, Rec. ITU-T Y.2060, 2012.

- [4] The National Intelligence Council, National Intelligence Council, and SRI Consulting Business Intelligence, Disruptive Civil Technologies: Six Technologies With Potential Impacts on US Interests Out to 2025. SRI Consulting Business Intelligence, Apr. 2008, p. 48.
- [5] D. Evans, "The Internet of Things—How the next evolution of the Internet is changing everything," CISCO, San Jose, CA, USA, White Paper, Apr. 2011.
- [6] R. Mehmood, F. Alam, N. N. Albogami, I. Katib, A. Albeshri, and S. M. Altowaijri, "UTiLearn: A personalised ubiquitous teaching and learning system for smart societies," *IEEE Access*, vol. 5, pp. 2615–2635, 2017.
- [7] H. B. Mitchell, Multi-Sensor Data Fusion: An Introduction. Berlin, Germany: Springer, 2007.
- [8] MIT Technology Review. (2013). 2013: The Year of the Internet of Things. [Online]. Available: https://www.technologyreview.com/ s/509546/2013-the-year-of-the-internet-of-things/
- [9] S. Li, L. Da Xu, and S. Zhao, "The Internet of Things: A survey," *Inf. Syst. Frontiers*, vol. 17, no. 2, pp. 243–259, 2015.
- [10] A. Al-Fuqaha, M. Guizani, M. Mohammadi, M. Aledhari, and M. Ayyash, "Internet of Things: A survey on enabling technologies, protocols, and applications," *IEEE Commun. Surveys Tuts.*, vol. 17, no. 4, pp. 2347–2376, 4th Quart., 2015.
- [11] J. Dong, D. Zhuang, Y. Huang, and J. Fu, "Advances in multi-sensor data fusion: Algorithms and applications," *Sensors*, vol. 9, no. 10, pp. 7771–7784, Sep. 2009.
- [12] F. Castanedo, "A review of data fusion techniques," Sci. World J., vol. 2013, Sep. 2013, Art. no. 704504.
- [13] B. Khaleghi, A. Khamis, F. O. Karray, and S. N. Razavi, "Multisensor data fusion: A review of the state-of-the-art," *Inf. Fusion*, vol. 14, no. 1, pp. 28–44, 2013.
- [14] N.-E. El Faouzi, H. Leung, and A. Kurian, "Data fusion in intelligent transportation systems: Progress and challenges—A survey," *Inf. Fusion*, vol. 12, no. 1, pp. 4–10, 2011.
- [15] I. M. Pires, N. M. Garcia, N. Pombo, and F. Flórez-Revuelta, "From data acquisition to data fusion: A comprehensive review and a roadmap for the identification of activities of daily living using mobile devices," *Sensors*, vol. 16, no. 2, p. 184, 2016.
- [16] X. Qin and Y. Gu, "Data fusion in the Internet of Things," *Procedia Eng.*, vol. 15, pp. 3023–3026, Dec. 2011.
- [17] S. Gite and H. Agrawal, "On context awareness for multisensor data fusion in IoT," in *Proc. 2nd Int. Conf. Comput. Commun. Technol.*, vol. 381. 2015, pp. 85–93.
- [18] M. Wang, C. Perera, P. P. Jayaraman, M. Zhang, P. Strazdins, and R. Ranjan, "City data fusion: Sensor data fusion in the Internet of Things," *Int. J. Distrib. Syst. Technol.*, vol. 7, no. 1, pp. 15–36, 2016.
- [19] D. L. Hall and J. Llinas, *Handbook of Multisensor Data Fusion*. New York, NY, USA: CRC Press, 2001, p. 529.
- [20] S. Vural and E. Ekici, "On multihop distances in wireless sensor networks with random node locations," *IEEE Trans. Mobile Comput.*, vol. 9, no. 4, pp. 540–552, Apr. 2010.
- [21] M. Kumar, D. P. Garg, and R. A. Zachery, "A generalized approach for inconsistency detection in data fusion from multiple sensors," in *Proc. Amer. Control Conf.*, 2006, pp. 2078–2083.
- [22] P. Smets, "Analyzing the combination of conflicting belief functions," *Inf. Fusion*, vol. 8, no. 4, pp. 387–412, 2007.
- [23] Data Fusion System Engineering Guidelines. USAF Space Warfare Center (SWC), 1997.
- [24] J. R. Raol, Multi-Sensor Data Fusion With MATLAB, 1st ed. Boca Raton, FL, USA: CRC Press, 2009.
- [25] D. P. Mandic et al., "Data fusion for modern engineering applications: An overview," in Artificial Neural Networks: Formal Models and Their Applications—ICANN (Lecture Notes in Computer Science). Heidelberg, Germany: Springer, 2005, p. 715.
- [26] T. E. Fortmann, Y. Bar-Shalom, and M. Scheffe, "Multi-target tracking using joint probabilistic data association," in *Proc. 19th IEEE Conf. Decision Control Including Symp. Adapt. Process.*, Dec. 1980, pp. 807–812.
- [27] S. H. Rezatofighi, A. Milan, Z. Zhang, Q. Shi, A. Dick, and I. Reid, "Joint probabilistic data association revisited," in *Proc. IEEE Int. Conf. Comput. Vis.*, Dec. 2015, pp. 3047–3055.
- [28] Z. Ding and L. Hong, "Bias phenomenon and compensation for PDA/JPDA algorithms," *Math. Comput. Model.*, vol. 27, no. 12, pp. 1–16, 1998.

- [29] S. Cong and L. Hong, "Computational complexity analysis for multiple hypothesis tracking," *Math. Comput. Model.*, vol. 29, no. 9, pp. 1–16, 1999.
- [30] S. Streltsov, "Airborne learning algorithms: A case of a ground target multiple hypothesis tracker," in *Proc. Int. Conf. Integr. Knowl. Intensive Multi-Agent Syst.*, Sep./Oct. 2003, pp. 190–195.
- [31] H. Holzapfel, K. Nickel, and R. Stiefelhagen, "Implementation and evaluation of a constraint-based multimodal fusion system for speech and 3D pointing gestures," in *Proc. 6th Int. Conf. Multimodal Interfaces (ICMI)*, 2004, pp. 175–182.
- [32] R. C. Luo, C.-C. Yih, and K. L. Su, "Multisensor fusion and integration: Approaches, applications, and future research directions," *IEEE Sensors J.*, vol. 2, no. 2, pp. 107–119, Apr. 2002.
- [33] M. Guerriero, L. Svensson, and P. Willett, "Bayesian data fusion for distributed target detection in sensor networks," *IEEE Trans. Signal Process.*, vol. 58, no. 6, pp. 3417–3421, Jun. 2010.
- [34] T. A. Biresaw, A. Cavallaro, and C. S. Regazzoni, "Tracker-level fusion for robust Bayesian visual tracking," *IEEE Trans. Circuits Syst. Video Technol.*, vol. 25, no. 5, pp. 776–789, May 2015.
- [35] L.-W. Fong, "An adaptive filter for multi-sensor track fusion," in Proc. Int. Conf. Signal Process. (ICSP), 2008, pp. 231–235.
- [36] L.-W. Fong, "Multi-sensor track fusion via multiple-model adaptive filter," in *Proc. IEEE Conf. Decision Control*, Dec. 2009, pp. 2327–2332.
- [37] L.-W. Fong, "Adaptive information matrix filtering fusion with nonlinear classifier," in *Proc. SICE Annu. Conf.*, 2010, pp. 2214–2219.
- [38] R. O. Duda, P. E. Hart, and D. G. Stork, *Pattern Classification*, 2nd ed. New York, NY, USA: Wiley, 2001.
- [39] L.-W. Fong, "Switching adaptive filter design using Bayesian classification approach for multi-sensor data fusion," in *Proc. 8th Asian Control Conf. (ASCC)*, 2011, pp. 1310–1315.
- [40] A. Noyvirt and R. Qiu, "Human detection and tracking in an assistive living service robot through multimodal data fusion," in *Proc. IEEE 10th Int. Conf. Ind. Inform.*, Jul. 2012, pp. 1176–1181.
- [41] S. Duffner and J.-M. Odobez, "Track creation and deletion framework for long-term online multiface tracking," *IEEE Trans. Image Process.*, vol. 22, no. 1, pp. 272–285, Jan. 2013.
- [42] D. L. Hall and S. A. H. McMullen, *Mathematical Techniques in Multi-Sensor Data Fusion*, 1st ed. Norwood, MA, USA: Artech House, 2004.
- [43] X. Zhu, F. Kui, and Y. Wang, "Predictive analytics by using Bayesian model averaging for large-scale Internet of Things," *Int. J. Distrib. Sensor Netw.*, vol. 2013, no. 3, pp. 1–10, 2013.
- [44] M. Kumar, D. P. Garg, and R. A. Zachery, "A method for judicious fusion of inconsistent multiple sensor data," *IEEE Sensors J.*, vol. 7, no. 5, pp. 723–733, May 2007.
- [45] Y. Guo, X. Cai, and J. Ma, "Improved JPDA for fast fault detection," in Proc. 30th Chin. Control Conf. (CCC), Jul. 2011, pp. 4167–4169.
- [46] J. Liu, C. Han, and Y. Hu, "Multiple target tracking using particle filter based multi-scan joint probabilistic data association filter," in *Proc. 30th Chin. Control Conf. (CCC)*, Jul. 2011, pp. 3134–3139.
- [47] J. Liu, C. Han, F. Han, and Y. Hu, "Multiple maneuvering target tracking by improved particle filter based on multiscan JPDA," *Math. Problems Eng.*, vol. 2012, Nov. 2012, Art. no. 372161.
- [48] G. Liu and G. Tian, "Distributed vision network for multiple target tracking using a dynamic hybrid consensus filter," in *Proc. IEEE Int. Conf. Autom. Sci. Eng.*, Nov. 2016, pp. 805–808.
- [49] J. Guo and R. Zhang, "Novel implementation of track-oriented multiple hypothesis tracking algorithm," *Chin. J. Electron.*, vol. 21, no. 4, pp. 770–774, 2012.
- [50] J. Hou, Y. Yang, Y. Chen, and L. Xu, "Multiple hypothesis tracker in the presence of RGPO/RGPI using amplitude information," in *Proc. 19th Int. Conf. Inf. Fusion*, 2016, pp. 703–710.
- [51] S. Coraluppi, C. Carthel, W. Kreamer, and A. Willsky, "Recent advances in multi-INT track fusion," in *Proc. IEEE Aerosp. Conf.*, Mar. 2016, pp. 1–13.
- [52] W. A. Abdulhafiz and A. Khamis, "Bayesian approach with pre- and postfiltering to handle data uncertainty and inconsistency in mobile robot local positioning," J. Intell. Syst., vol. 23, no. 2, pp. 133–154, 2014.
- [53] F. Bourgault *et al.*, "Scalable Bayesian human-robot cooperation in mobile sensor networks," in *Proc. IEEE/RSJ Int. Conf. Intell. Robot. Syst. (IROS)*, Sep. 2008, pp. 2342–2349.
- [54] W. C. Chen and N. V. Z. Shuley, "Robust target identification using a modified generalized likelihood ratio test," *IEEE Trans. Antennas Propag.*, vol. 62, no. 1, pp. 264–273, Jan. 2014.

- [55] J. Liu, H. Li, and B. Himed, "Two target detection algorithms for passive multistatic radar," *IEEE Trans. Signal Process.*, vol. 62, no. 22, pp. 5930–5939, Nov. 2014.
- [56] B. Wang. (2013). Next big future: X-Apple Siri Director Samsung's AI for the Internet of Things. Next Big Future Inc., accessed on Dec. 22, 2016. [Online]. Available: http://www.nextbigfuture.com/2013/11/x-apple-siridirector-is-taking.html
- [57] P. Mishra. (2015). Google is making a land grab for the Internet of Things. Tech Crunch, accessed on Dec. 20, 2016. [Online]. Available: https://techcrunch.com/2014/01/27/google-is-making-a-land-grabfor-the-internet-of-things/
- [58] BI-Intelligence. (2016). The Google of China announces new AI and Internet of Things initiatives. Business Insider, accessed on Dec. 20, 2016. [Online]. Available: http://www.businessinsider.com/baidu-announcesnew-ai-and-internet-of-things-initiatives-2016-9
- [59] S. B. Kotsiantis, "Supervised machine learning: A review of classification techniques," *Informatica*, vol. 31, no. 1, pp. 249–268, 2007.
- [60] F. Chen, P. Deng, J. Wan, D. Zhang, A. V. Vasilakos, and X. Rong, "Data mining for the Internet of Things: Literature review and challenges," *Int. J. Distrib. Sensor Netw.*, vol. 11, no. 8, pp. 249–268, 2015.
- [61] C. C. Aggarwal, N. Ashish, and A. Sheth, "The Internet of Things: A survey from the data-centric perspective," in *Managing and Mining Sensor Data*. New York, NY, USA: Springer, 2013, pp. 383–428.
- [62] S. Bin, L. Yuan, and W. Xiaoyi, "Research on data mining models for the Internet of Things," in *Proc. Int. Conf. Image Anal. Signal Process.*, 2010, pp. 127–132.
- [63] C.-W. Tsai, C.-F. Lai, M.-C. Chiang, and L. T. Yang, "Data mining for Internet of Things: A survey," *IEEE Commun. Surveys Tuts.*, vol. 16, no. 1, pp. 77–97, 1st Quart., 2014.
- [64] S. Challa, M. Palaniswami, and A. Shilton, "Distributed data fusion using support vector machines," in *Proc. 5th Int. Conf. Inf. Fusion (FUSION)* (*IEEE Cat.No.02EX5997*), vol. 1. Jul. 2002 pp. 881–885.
- [65] S. Zhao, "Remote sensing data fusion using support vector machine," in Proc. IEEE Int. Geosci. Remote Sens. Symp., vol. 4. Sep. 2004, pp. 2575–2578.
- [66] B. Bigdeli, F. Samadzadegan, and P. Reinartz, "A decision fusion method based on multiple support vector machine system for fusion of hyperspectral and LIDAR data," *Int. J. Image Data Fusion*, vol. 5, no. 3, pp. 196–209, 2014.
- [67] A. Merentitis, C. Debes, R. Heremans, and N. Frangiadakis, "Automatic fusion and classification of hyperspectral and LiDAR data using random forests," in *Proc. IEEE Int. Geosci. Remote Sens. Symp.*, Jul. 2014, pp. 1245–1248.
- [68] E. T. McGee and J. D. McGregor, "Using dynamic adaptive systems in safety-critical domains," in *Proc. IEEE/ACM 11th Int. Symp. Softw. Eng. Adapt. Self-Manag. Syst.*, May 2016, pp. 115–121.
- [69] T. P. Banerjee and S. Das, "Multi-sensor data fusion using support vector machine for motor fault detection," *Inf. Sci.*, vol. 217, pp. 96–107, Dec. 2012.
- [70] A. Starzacher and B. Rinner, "Evaluating KNN, LDA and QDA classification for embedded online feature fusion," in *Proc. Int. Conf. Intell. Sensors, Sens. Netw. Inf. Process. (ISSNIP)*, 2008, pp. 85–90.
- [71] N. El Gayar, F. Schwenker, and C. Y. Suen, Eds., Artificial Neural Networks in Pattern Recognition (Lecture Notes in Artificial Intelligence), vol. 8774. Switzerland: Springer International Publishing, 2014.
- [72] D. Chen, K. Zhang, and T. Liao, "Practical travel time prediction algorithms based on neural network and data fusion for urban expressway," in *Proc. 6th Int. Conf. Natural Comput. (ICNC)*, vol. 4. 2010, pp. 1754–1758.
- [73] H. B. Azad, S. Mekhilef, and V. G. Ganapathy, "Long-term wind speed forecasting and general pattern recognition using neural networks," *IEEE Trans. Sustain. Energy*, vol. 5, no. 2, pp. 546–553, Apr. 2014.
- [74] F. Zhang, K. Cho, J. Choi, Y.-M. Lee, and K. Y. Lee, "A study on wind speed prediction using artificial neural network at Jeju Island in Korea II," in *Proc. IEEE 54th Int. Midwest Symp. Circuits Syst.*, Aug. 2011, pp. 1–4.
- [75] P. M. Fonte, G. X. Silva, and J. C. Quadrado, "Wind speed prediction using artificial neural networks," *WSEAS Trans. Syst.*, vol. 4, no. 4, pp. 379–384, 2005.
- [76] Z. Liu, W. Gao, Y.-H. Wan, and E. Muljadi, "Wind power plant prediction by using neural networks," in *Proc. IEEE Energy Convers. Congr. Expo.*, Sep. 2012, pp. 3154–3160.
- [77] M. Altini, D. Brunelli, E. Farella, and L. Benini, "Bluetooth indoor localization with multiple neural networks," in *Proc. IEEE 5th Int. Symp. Wireless Pervasive Comput.*, May 2010, pp. 295–300.

- [78] H. Singh et al., "Real-life applications of fuzzy logic," Adv. Fuzzy Syst., vol. 2013, Apr. 2013, Art. no. 581879.
- [79] P. Manjunatha, A. K. Verma, and A. Srividya, "Multi-sensor data fusion in cluster based wireless sensor networks using fuzzy logic method," in *Proc. IEEE Region 10 3rd Int. Conf. Ind. Inf. Syst.*, Dec. 2008, pp. 1–6.
- [80] C. Lee, Y. Han, S. Jeon, D. Seo, and I. Jung, "Smart parking system for Internet of Things," in *Proc. IEEE Int. Conf. Consum. Electron. (ICCE)*, Jan. 2016, pp. 263–264.
- [81] A. Khanna and R. Anand, "IoT based smart parking system," in Proc. Int. Conf. Internet Things Appl. (IOTA), 2016, pp. 266–270.
- [82] Z. Jian, C. Hongbing, S. Jie, and L. Haitao, "Data fusion for magnetic sensor based on fuzzy logic theory," in *Proc. 4th Int. Conf. Intell. Comput. Technol. Autom.*, Mar. 2011, pp. 87–92.
- [83] I. Mobin, M. Abid-Ar-Rafi, N. Islam, and R. Hasan, "An intelligent fire detection and mitigation system safe from fire (SFF)," *Int. J. Comput. Appl.*, vol. 133, no. 6, pp. 1–7, 2016.
- [84] A. M. Aziz, M. Tummala, and R. Cristi, "Fuzzy logic data correlation approach in multisensor Umultitarget tracking systems," *Signal Process.*, vol. 76, no. 2, pp. 195–209, 1999.
- [85] F. En, X. Wei-Xin, L. Zong-Xiang, and L. Peng-Fei, "A novel fuzzy trackto-track association algorithm in distributed sensor network," *J. Signal Process.*, vol. 27, no. 10, pp. 1561–1565, 2011.
- [86] A. M. Aziz, "A new fuzzy clustering approach for data association and track fusion in multisensor-multitarget environment," in *Proc. Aerosp. Conf.*, 2011, pp. 1–10.
- [87] G.-L. Ran and H.-H. Wu, "Fuzzy fusion approach for object tracking," in *Proc. 2nd WRI Global Congr. Intell. Syst. (GCIS)*, vol. 3. 2010, pp. 219–222.
- [88] W. Du, H. Ning, Y. Wei, and J. Wang, "Fuzzy double-threshold track association algorithm using adaptive threshold in distributed multisensormultitarget tracking systems," in Proc. IEEE Int. Conf. Green Comput. Commun. IEEE Internet Things IEEE Cyber, Phys. Soc. Comput. (GreenCom-iThings-CPSCom), Aug. 2013, pp. 1133–1137.
- [89] N. Mort and P. J. Escamilla-Ambrosio, "Hybrid Kalman filter-fuzzy logic adaptive multisensor data fusion architectures," in *Proc. 42nd IEEE Conf. Decision Control*, Dec. 2003, pp. 5215–5220.
- [90] V. Subramanian, T. F. Burks, and W. E. Dixon, "Sensor fusion using fuzzy logic enhanced Kalman filter for autonomous vehicle guidance in citrus groves," *Trans. ASABE*, vol. 52, no. 5, pp. 1411–1422, 2009.
- [91] G. Shafer, A Mathematical Theory of Evidence, 1st ed. Princeton, NJ, USA: Princeton Univ. Press, 1976.
- [92] A. Bellenger and S. Gatepaille. (2011). "Uncertainty in ontologies: Dempster–Shafer theory for data fusion applications." [Online]. Available: https://arxiv.org/abs/1106.3876
- [93] F. Rottensteiner, J. Trinder, S. Clode, and K. Kubik, "Using the Dempster–Shafer method for the fusion of LIDAR data and multi-spectral images for building detection," *Inf. Fusion*, vol. 6, no. 4, pp. 283–300, 2005.
- [94] D. Jiang, D. Zhuang, Y. Huang, and J. Fu, "Survey of multispectral image fusion techniques in remote sensing applications," in *Image Fusion and Its Applications*. Croatia: INTECH, 2011.
- [95] V. Saeidi, B. Pradhan, M. O. Idrees, and Z. A. Latif, "Fusion of airborne LiDAR with multispectral SPOT 5 image for enhancement of feature extraction using Dempster–Shafer theory," *IEEE Trans. Geosci. Remote Sens.*, vol. 52, no. 10, pp. 6017–6025, Oct. 2014.
- [96] H. Janssen and W. Nichsen, "Vehicle surround sensing based on information fusion of monocular video and digital map," in *Proc. IEEE Intell. Veh. Symp.*, Jun. 2004, pp. 244–249.
- [97] S. K. Gehrig, S. Wagner, and U. Franke, "System architecture for an intersection assistant fusing image, map, and GPS information," in *Proc. IEEE Intell. Veh. Symp.*, Jun. 2003, pp. 144–149.
- [98] D. Nienhuser, T. Gumpp, and J. M. Zollner, "A situation context aware Dempster–Shafer fusion of digital maps and a road sign recognition system," in *Proc. IEEE Intell. Veh. Symp.*, Jun. 2009, pp. 1401–1406.
- [99] H. Jamshidi, T. Lukaszewicz, A. Kashi, A. Berghuvud, H.-J. Zepernick, and S. Khatibi, "Fusion of digital map traffic signs and camera-detected signs," in *Proc. 5th Int. Conf. Signal Process. Commun. Syst. (ICSPCS)*, 2011, pp. 1–7.
- [100] R. Martin, J. Zhang, and C. Liu, "Dempster–Shafer theory and statistical inference with weak beliefs," *Statist. Sci.*, vol. 25, no. 1, pp. 72–87, 2010.
 [101] S. Challa and D. Koks, "Bayesian and Dempster–Shafer fusion,"
- [101] S. Challa and D. Koks, "Bayesian and Dempster–Shafer fusion," Sadhana, vol. 29, no. 2, pp. 145–176, Apr. 2004.
- [102] T. W. Martin and K. Chang, "A data fusion formulation for decentralized estimation predictions under communications uncertainty," in *Proc. 9th Int. Conf. Inf. Fusion (FUSION)*, 2006, pp. 1–7.

- [103] A. Abdelgawad, "Distributed data fusion algorithm for wireless sensor network," in *Proc. 11th IEEE Int. Conf. Netw., Sens. Control*, Apr. 2014, pp. 334–337.
- [104] M. Dhivya, M. Sundarambal, and L. N. Anand, "Energy efficient computation of data fusion in wireless sensor networks using cuckoo based particle approach (CBPA)," *Int. J. Commun. Netw. Syst. Sci.*, vol. 4, pp. 249–255, Apr. 2011.
- [105] Z. Lu, S. L. Tan, and J. Biswas, "Function placement of data fusion for active networks paradigm in wireless sensor networks," in *Proc.*-8th IEEE Int. Conf. Mobile Ad-Hoc Sens. Syst. (MASS), Oct. 2011, pp. 152–154.
- [106] K. Zhang, C. Li, and W. Zhang, "Wireless sensor data fusion algorithm based on the sensor scheduling and batch estimate," *Int. J. Future Comput. Commun.*, vol. 2, no. 4, p. 333, Aug. 2013.
- [107] V. Sasikala and C. Ch, "Energy efficient multipath data fusion technique for wireless sensor networks," ACEEE Int. J. Netw. Secur., vol. 3, no. 2, pp. 34–41, 2012.
- [108] Y. Huang, Energy Efficient Distributed Data Fusion in Multihop Wireless Sensor Networks. Riverside, CA, USA: Univ. California Riverside, 2010.
- [109] K. C. Chang, C. Y. Chong, and S. Mori, "Analytical and computational evaluation of scalable distributed fusion algorithms," *IEEE Trans. Aerosp. Electron. Syst.*, vol. 46, no. 4, pp. 2022–2034, Oct. 2010.
- [110] J. Choo, S. Bohn, G. C. Nakamura, A. M. White, and H. Park, "Heterogeneous data fusion via space alignment using nonmetric multidimensional scaling," in *Proc. SIAM Int. Conf. Data Mining*, 2012, pp. 177–188.
- [111] Y. S. Chiou and F. Tsai, "A reduced-complexity data-fusion algorithm using belief propagation for location tracking in heterogeneous observations," *IEEE Trans. Cybern.*, vol. 44, no. 6, pp. 922–935, Jun. 2014.
- [112] E. F. Lock and D. B. Dunson, "Bayesian consensus clustering," *Bioin-formatics*, vol. 29, no. 20, pp. 2610–2616, 2013.
- [113] P. Ray, L. Zheng, J. Lucas, and L. Carin, "Bayesian joint analysis of heterogeneous genomics data," *Bioinformatics*, vol. 30, no. 10, pp. 1370–1376, 2014.
- [114] I. González, S. Déjean, P. G. P. Martin, O. Gonçalves, P. Besse, and A. Baccini, "Highlighting relationships between heterogeneous biological data through graphical displays based on regularized canonical correlation analysis," *J. Biol. Syst.*, vol. 17, no. 2, pp. 173–199, 2009.
- [115] H. Hotelling, "Relations between two sets of variates," *Biometrika*, vol. 28, nos. 3–4, pp. 321–377, 1936.
- [116] K.-A. L. Cao, P. G. Martin, C. Robert-Granié, and P. Besse, "Sparse canonical methods for biological data integration: Application to a crossplatform study," *BMC Bioinformatics*, vol. 10, no. 1, p. 34, 2009.
- [117] H. Wold, Estimation of Principal Components and Related Models by Iterative Least Squares. New York, NY, USA: Academic, 1966.
- [118] N. Alaydie, F. Fotouhi, C. K. Reddy, and H. Soltanian-Zadeh, "Noise and outlier filtering in heterogeneous medical data sources," in *Proc. Workshop Database Expert Syst. Appl.*, 2010, pp. 115–119.
- [119] L. Fricoteaux, I. M. Thouvenin, and J. Olive, "Heterogeneous data fusion for an adaptive training in informed virtual environment," in *Proc. IEEE Int. Conf. Virtual Environ. Human-Comput. Inter. Meas. Syst. (VECIMS)*, Sep. 2011, pp. 113–118.
- [120] J. Zhao, Y. Liu, Y. Cheng, Y. Qiang, and X. Zhang, "Multisensor data fusion for wildfire warning," in *Proc. 10th Int. Conf. Mobile Ad-Hoc Sens. Netw.*, Dec. 2014, pp. 46–53.
- [121] H. Zhang, T. S. Huang, N. M. Nasrabadi, and Y. Zhang, "Heterogeneous multi-metric learning for multi-sensor fusion," in *Proc. 14th Int. Conf. Inf. Fusion*, 2011, pp. 1–8.
- [122] P. Pong, M. Morelande, and S. Challa, "Heterogeneous fusion with a combined evidential, probability and OWA methods for target classification," in *Proc. 13th Conf. Inf. Fusion*, Jul. 2010, pp. 1–12.
- [123] U. Rashid, H. D. Tuan, P. Apkarian, and H. H. Kha, "Multisensor data fusion in nonlinear Bayesian filtering," in *Proc. 4th Int. Conf. Commun. Electron.*, Aug. 2012, pp. 351–354.
- [124] U. Rashid, H. D. Tuan, H. H. Kha, and H. H. Nguyen, "Semi-definite programming for distributed tracking of dynamic objects by nonlinear sensor network," in *Proc. IEEE Int. Conf. Acoust. Speech Signal Process. (ICASSP)*, May 2011, pp. 3532–3535.
- [125] R. Zhan and J. Wan, "Iterated unscented Kalman filter for passive target tracking," *IEEE Trans. Aerosp. Electron. Syst.*, vol. 43, no. 3, pp. 1155–1163, Jul. 2007.
- [126] S. J. Julier and J. K. Uhlmann, "A new extension of the Kalman filter to nonlinear systems," *Proc. AeroSense 11th Int. Symp. Aerospace/Defence Sens., Simulation Control*, 1997, pp. 1–2.

- [127] S. Julier, J. Uhlmann, and H. F. Durrant-Whyte, "A new method for the nonlinear transformation of means and covariances in filters and estimators," *IEEE Trans. Autom. Control*, vol. 45, no. 3, pp. 477–482, Mar. 2000.
- [128] D. L. Hall and J. Llinas, "An introduction to multisensor data fusion," *Proc. IEEE*, vol. 85, no. 1, pp. 6–23, Jan. 1997.
 [120] S. A. Parke, H. P. T.
- [129] S. A. Pasha, H. D. Tuan, and B. N. Vo, "Nonlinear Bayesian filtering using the unscented linear fractional transformation model," *IEEE Trans. Signal Process.*, vol. 58, no. 2, pp. 477–489, Feb. 2010.
- [130] M. A. A. Akhoundi and E. Valavi. (2010). "Multi-sensor fuzzy data fusion using sensors with different characteristics." [Online]. Available: https://arxiv.org/abs/1010.6096
- [131] C. Wang, H. Liu, and Y. Gao, "Scene-adaptive hierarchical data association for multiple objects tracking," *IEEE Signal Process. Lett.*, vol. 21, no. 6, pp. 697–701, Jun. 2014.
- [132] Y. Fu, Q. Ling, and Z. Tian, "Distributed sensor allocation for multitarget tracking in wireless sensor networks," *IEEE Trans. Aerosp. Electron. Syst.*, vol. 48, no. 4, pp. 3538–3553, Oct. 2012.
- [133] F. Tango, E. Richter, U. Scheunert, and G. Wanielik, "Advanced multiple objects tracking by fusing radar and image sensor data—Application on a case study," in *Proc. 11th Int. Conf. Inf. Fusion*, Jul. 2008, pp. 1–7.
- [134] H. Talebi and A. M. A. Hemmatyar, "Asynchronous track-to-track fusion by direct estimation of time of sample in sensor networks," *IEEE Sens. J.*, vol. 14, no. 1, pp. 210–217, Jan. 2014.
- [135] J. Kichun, C. Keounyup, and S. Myoungho, "Interacting multiple model filter-based sensor fusion of GPS with in-vehicle sensors for real-time vehicle positioning," *IEEE Trans. Intell. Transp. Syst.*, vol. 13, no. 1, pp. 329–343, Mar. 2012.
- [136] X. Chang et al., "Sensor placement algorithms for fusion-based surveillance networks," *IEEE Trans. Parallel Distrib. Syst.*, vol. 22, no. 8, pp. 1407–1414, Aug. 2011.
- [137] Y. Han, H. Zhu, and C. Han, "Track-to-track association in the presence of sensor bias and the relative bias estimation," in *Proc. 15th Int. Conf. Inf. Fusion (FUSION)*, 2012, pp. 1044–1050.
- [138] K. Weiss, D. Stueker, and A. Kirchner, "Target modeling and dynamic classification for adaptive sensor data fusion," in *Proc. IEEE Intell. Veh. Symp.*, Jun. 2003, pp. 132–137.
- [139] S. Matzka and R. Altendorfer, "A comparison of track-to-track fusion algorithms for automotive sensor fusion," in *Proc. IEEE Int. Conf. Multisens. Fusion Integr. Intell. Syst. (MFI)*, vol. 35. Aug. 2008, pp. 69–81.
- [140] Y. Bar-Shalom, "On the track-to-track correlation problem," *IEEE Trans. Autom. Control*, vol. 26, no. 2, pp. 571–572, Apr. 1981.
- [141] J. K. Uhlmann, "Covariance consistency methods for fault-tolerant distributed data fusion," *Inf. Fusion*, vol. 4, no. 3, pp. 201–215, Sep. 2003.
- [142] Statlect. (2014). Cross Covariance Matrix. [Online]. Available: http://www.statlect.com
- [143] M. Aeberhard, S. Schlichtharle, N. Kaempchen, and T. Bertram, "Track-to-track fusion with asynchronous sensors using information matrix fusion for surround environment perception," *IEEE Trans. Intell. Transp. Syst.*, vol. 13, no. 4, pp. 1717–1726, Dec. 2012.
- [144] BBC Asia. (2014). How do you Track a Plane?. [Online]. Available: http://www.bbc.com/news/world-asia-pacific-26544554
- [145] A. Yilmaz, O. Javed, and M. Shah, "Object tracking: A survey," ACM Comput. Surv., vol. 38, no. 4, p. 13, 2006.
- [146] P. Xu, F. Davoine, H. Zhao, and T. Denœux, "Multimodal information fusion for urban scene understanding," *Mach. Vis. Appl.*, vol. 27, no. 3, pp. 331–349, 2014.
- [147] D. Nuss, M. Thom, A. Danzer, and K. Dietmayer, "Fusion of laser and monocular camera data in object grid maps for vehicle environment perception," in *Proc. 17th Int. Conf. Inf. Fusion*, Jul. 2014, pp. 1–8.
- [148] A. Tsukada and A. Background, "Road structure based scene understanding for Intelligent Vehicle Systems," in *Proc. IEEE/RSJ Int. Conf. Intell. Robot. Syst.*, Oct. 2010, pp. 5557–5562.
- [149] Q. Hu, P. Wang, C. Shen, and F. Porikli, "Pushing the limits of deep CNNs for pedestrian detection," in *Proc. Comput. Vis. Pattern Recognit.*, 2016, pp. 123–144.
- [150] C. Premebida, J. Batista, and U. Nunes, "Pedestrian detection combining RGB and dense LIDAR data," in *Proc. IEEE/RSJ Int. Conf. Intell. Robot. Syst. (IROS)*, Sep. 2014, pp. 4112–4117.
- [151] H. Cho, Y. Seo, B. V. K. V. Kumar, and R. R. Rajkumar, "A multi-sensor fusion system for moving object detection and tracking in urban driving environments," in *Proc. IEEE Int. Conf. Robot. Autom.*, Jun. 2014, pp. 1836–1843.
- [152] N. Chumerin and M. M. van Hulle, "Cue and sensor fusion for independent moving objects detection and description in driving scenes," in Signal Processing Techniques for Knowledge Extraction and Information Fusion. Springer, LLC, 2008, pp. 161–180.

- [153] C. Häne, T. Sattler, and M. Pollefeys, "Obstacle detection for selfdriving cars using only monocular cameras and wheel odometry," in *Proc. IEEE/RSJ Int. Conf. Intell. Robot. Syst.*, Oct. 2015, pp. 5101–5108.
- [154] Y. Zhao, J. Li, L. Li, M. Zhang, and L. Guo, "Environmental perception and sensor data fusion for unmanned ground vehicle," *Math. Probl. Eng.*, vol. 2013, Oct. 2013, Art. no. 903951.
- [155] A. S. Huang, "Lane estimation for autonomous vehicles using vision and LIDAR," M.S. thesis, Massachusetts Inst. Technol., Feb. 2010, pp. 1–114.
- [156] D. Z. Morris, "Google self-driving car in serious crash in Mountain View," Fortune, 2016.
- [157] S. Levin and N. Woolf, "Tesla driver killed while using autopilot was watching Harry Potter, witness says," *Guardian*, 2016.
- [158] H. P. Levy. (2015). Top 10 Technology Trends Signal the Digital Mesh, Gartner. [Online]. Available: http://www.gartner.com/ smarterwithgartner/top-ten-technology-trends-signal-the-digital-mesh/
- [159] Y. LeCun, Y. Bengio, and G. Hinton, "Deep learning," *Nature*, vol. 521, no. 7553, pp. 436–444, May 2015.
- [160] H. Wang and B. Raj, "A survey: Time travel in deep learning space: An introduction to deep learning models and how deep learning models evolved from the initial ideas," *Cornell Univ. Library*, pp. 1–43, Nov. 2015.
- [161] L. Deng, "A tutorial survey of architectures, algorithms, and applications for deep learning," *APSIPA Trans. Signal Inf. Process.*, vol. 3, no. e2, pp. 1–29, Jan. 2014.
 [162] Y. Bengio, "Learning deep architectures for AI," *Found. Trends Mach.*
- [162] Y. Bengio, "Learning deep architectures for AI," Found. Trends Mach. Learn., vol. 2, no. 1, pp. 1–127, 2009.
- [163] J. Ngiam, A. Khosla, M. Kim, J. Nam, H. Lee, and A. Y. Ng, "Multimodal deep learning," in Proc. 28th Int. Conf. Mach. Learn., 2011, pp. 689–696.
- [164] H. Martínez and G. Yannakakis, "Deep multimodal fusion: Combining discrete events and continuous signals," in *Proc. 16th Int. Conf. Multimodal Interact.*, 2014, pp. 34–41.
- [165] N. Audebert, B. Le Saux, and S. Lefèvre. (Sep. 2016). "Semantic segmentation of earth observation data using multimodal and multi-scale deep networks." [Online]. Available: https://arxiv.org/abs/1609.06846
- [166] L. Zhang and H. Gao, "A deep learning-based multi-sensor data fusion method for degradation monitoring of ball screws," in *Proc. Prognostics Syst. Health Manage. Conf. (PHM-Chengdu)*, Oct. 2016, pp. 1–6.
- [167] P. O'Connor, D. Neil, S. C. Liu, T. Delbruck, and M. Pfeiffer, "Real-time classification and sensor fusion with a spiking deep belief network," *Front. Neurosci.*, vol. 7, no. 7, pp. 1–13, Oct. 2013.
- [168] E. Park, X. Han, T. L. Berg, and A. C. Berg, "Combining multiple sources of knowledge in deep CNNs for action recognition," *J. Chem. Inf. Model.*, vol. 53, no. 9, pp. 1689–1699, 2013.
- [169] X. Tao, D. Kong, Y. Wei, and Y. Wang, "A big network traffic data fusion approach based on fisher and deep auto-encoder," *Information*, vol. 7, no. 2, p. 20, 2016.
- [170] D. Neil and S.-C. Liu, "Effective sensor fusion with event-based sensors and deep network architectures," in *Proc. IEEE Int. Symp. Circuits Syst.*, May 2016, pp. 2282–2285.
- [171] J. Wagner, V. Fischer, M. Herman, and S. Behnke, "Multispectral pedestrian detection using deep fusion convolutional neural networks," in *Proc. 24th Eur. Symp. Artif. Neural Netw., Comput. Intell. Mach. Learn.*, Apr. 2016, pp. 509–514.
- [172] A. Zanella, N. Bui, A. Castellani, L. Vangelista, and M. Zorzi, "Internet of Things for smart cities," *IEEE Internet Things J.*, vol. 1, no. 1, pp. 22–32, Feb. 2014.
- [173] H. Arasteh *et al.*, "Iot-based smart cities: A survey," in *Proc. IEEE 16th Int. Conf. Environ. Elect. Eng. (EEEIC).* Jun. 2016, pp. 1–6.
 [174] A. J. Jara, D. Genoud, and Y. Bocchi, "Big data for smart cities with
- [174] A. J. Jara, D. Genoud, and Y. Bocchi, "Big data for smart cities with KNIME a real experience in the SmartSantander testbed," *Softw.-Pract. Exp.*, vol. 45, no. 8, pp. 1145–1160, Aug. 2015.
- [175] K. Pollhammer, T. Novak, P. Raich, W. Kastner, A. Treytl, and G. Kovacs, "Open traffic data platform for scenario-based control," in *Proc. 42nd Annu. Conf. IEEE Ind. Electron. Soc. (IECON)*, Oct. 2016, pp. 4677–4682.
- [176] A. Kalyanaraman and K. Whitehouse, "An event-based data fusion algorithm for smart cities," in *Proc. Adjunct ACM Int. Joint Conf. Pervasive Ubiquitous Comput., Int. Symp. Wearable Comput.*, 2015, pp. 1575–1582.
- [177] J. Rivera-Mejía, C. Seáñez-Hernández, A. Hernández-López, and A. López-Pérez, "Intelligent sensor with data fusion to improve the care and management of water," in *Proc. IEEE Int. Instrum. Meas. Technol. Conf.*, May 2011, pp. 26–30.
- [178] J. Rivera-Mejía, J. Rivera-Jacquez, C. Modesto-Acosta, and M. Tena-Vega, "Auto-compensed smart sensor with F-ISE to continuous monitoring in public water supplies," in *Proc. IEEE Int. Instrum. Meas. Technol. Conf. (I2MTC)*, May 2012, pp. 2628–2633.
- VOLUME 5, 2017

- [179] N.-B. Chang and S. Imen, "Multi-sensor acquisition, data fusion, criteria mining and alarm triggering for decision support in urban water infrastructure systems," in *Proc. IEEE Int. Conf. Syst., Man, Cybern.* (SMC), Oct. 2015, pp. 539–544.
- [180] G. Bello-Orgaz, J. J. Jung, and D. Camacho, "Social big data: Recent achievements and new challenges," *Inf. Fusion*, vol. 28, pp. 45–59, Mar. 2016.
- [181] X. Z. Wang, X. L. Bi, Z. Q. Ge, and L. Li, "Deep data fusion model for risk perception and coordinated control of smart grid," in *Proc. Int. Conf. Estim. Detect. Inf. Fusion (ICEDIF)*, Jan. 2015, pp. 110–113.
- [182] S. Li and X. Wang, "Cooperative change detection for voltage quality monitoring in smart grids," *IEEE Trans. Inf. Forensics Security*, vol. 11, no. 1, pp. 86–99, Jan. 2016.
- [183] S. P. Talebi, S. Kanna, Y. Xia, and D. P. Mandic, "A distributed quaternion Kalman filter with applications to fly-by-wire systems," in *Proc. IEEE Int. Conf. Digit. Signal Process. (DSP)*, Oct. 2016, pp. 30–34.
- [184] X. Sevillano, E. Màrmol, and V. Fernandez-Arguedas, "Towards smart traffic management systems: Vacant on-street parking spot detection based on video analytics," in *Proc. 17th Int. Conf. Inf. Fusion (FUSION)*, Jul. 2014, pp. 1–8.
- [185] Y. Wang, B. Liang, W. Zheng, L. Huang, and H. Liu, "The development of a smart taxicab scheduling system: A multi-source data fusion perspective," in *Proc. IEEE 16th Int. Conf. Data Mining (ICDM)*, Dec. 2016, pp. 1275–1280.
- [186] L. Yu, D. Shao, and H. Wu, "Next generation of journey planner in a smart city," *Proc. IEEE Int. Conf. Data Mining Workshop (ICDMW)*, Nov. 2015, pp. 422–429.
- [187] P. J. F. White, B. W. Podaima, and M. R. Friesen, "Algorithms for smartphone and tablet image analysis for healthcare applications," *IEEE Access*, vol. 2, pp. 831–840, Aug. 2014.
- [188] D. De, P. Bharti, S. K. Das, and S. Chellappan, "Multimodal wearable sensing for fine-grained activity recognition in healthcare," *IEEE Internet Comput.*, vol. 19, no. 5, pp. 26–35, Sep./Oct. 2015.
- [189] Y. Chen, W. Gao, S. Wang, and S. Jiao, "Demo abstract: WearableHUB: An open pervasive wearable data fusion platform for personal health management," in *Proc. IEEE Int. Conf. Pervasive Comput. Commun. Workshops (PerCom Workshops)*, Mar. 2015, pp. 217–219.
- [190] Z. Alazawi, O. Alani, M. B. Abdljabar, S. Altowaijri, and R. Mehmood, "A smart disaster management system for future cities," in *Proc. ACM Int. Workshop Wireless Mobile Technol. Smart Cities (WiMobCity)*, 2014, pp. 1–10.
 [191] Z. Alazawi, M. B. Abdljabar, S. Altowaijri, A. M. Vegni, and
- [191] Z. Alazawi, M. B. Abdljabar, S. Altowaijri, A. M. Vegni, and R. Mehmood, *ICDMS: An Intelligent Cloud Based Disaster Management System for Vehicular Networks* (Lecture Notes in Computer Science), vol. 7266. Vilnius, Lithuania: Springer, 2012.
- [192] G. Ayres and R. Mehmood, "On discovering road traffic information using virtual reality simulations," in *Proc. 11th Int. Conf. Comput. Modelling Simulation*, Mar. 2009, pp. 411–416.
- [193] J. Schlingensiepen, F. Nemtanu, R. Mehmood, and L. McCluskey, "Autonomic transport management systems—Enabler for smart cities, personalized medicine, participation and industry grid/industry 4.0," in *Intelligent Transportation Systems—Problems and Perspectives*, A. Sladkowski and W. Pamula, Eds. London, U.K.: Springer, 2016, pp. 3–35.
- [194] J. Schlingensiepen, R. Mehmood, and F. C. Nemtanu, "Framework for an autonomic transport system in smart cities," *Cybern. Inf. Technol.*, vol. 15, no. 5, pp. 50–62, 2015.
- [195] R. Mehmood and G. Graham, "Big data logistics: A health-care transport capacity sharing model," *Procedia Comput. Sci.*, vol. 64, pp. 1107–1114, 2015.
- [196] R. Mehmood and J. A. Lu, "Computational Markovian analysis of large systems," J. Manuf. Technol. Manag., vol. 22, no. 6, pp. 804–817, 2011.
- [197] R. Mehmood, R. Meriton, G. Graham, P. Hennelly, and M. Kumar, "Exploring the influence of big data on city transport operations: A Markovian approach," *Int. J. Oper. Prod. Manag.*, vol. 37, no. 1, pp. 75–104, 2016.
- [198] G. Graham, R. Mehmood, and E. Coles, "Exploring future cityscapes through urban logistics prototyping: A technical viewpoint," *Supply Chain Manage., Int. J.*, vol. 20, no. 3, pp. 341–352, 2015.
- [199] G. Ayres and R. Mehmood, LocPriS: A Security and Privacy Preserving Location Based Services Development Framework (Lecture Notes in Artificial Intelligence), vol. 6279. Heidelberg, Germany: Springer, 2010.
- [200] R. Mehmood, R. Alturki, and S. Zeadally, "Multimedia applications over metropolitan area networks (MANs)," *J. Netw. Comput. Appl.*, vol. 34, no. 5, pp. 1518–1529, Sep. 2011.

[201] R. Mehmood and R. Alturki, "A scalable multimedia qos architecture for ad hoc networks," *Multimed. Tools Appl.*, vol. 54, no. 3, pp. 551–568, 2011.

IEEE Access

- [202] R. Mehmood, M. A. Faisal, and S. Altowaijri, "Future Networked Healthcare Systems: A Review and Case Study," in *Big Data: Concepts, Methodologies, Tools, Application, I. Management Association.* Hershey, PA, USA: IGI Global, 2016, pp. 2429–2457.
- [203] R. Alturki, K. Nwizege, R. Mehmood, and M. Faisal, "End to end wireless multimedia service modelling over a metropolitan area network," in *Proc. 11th Int. Conf. Comput. Modelling Simulation* (UKSim), Mar. 2009, pp. 532–537.
- [204] G. Graham and R. Mehmood, "The strategic prototype 'crime-sourcing' and the science/science fiction behind it," *Technol. Forecast. Soc. Change*, vol. 84, pp. 86–92, 2014.
- [205] R. Mehmood and M. Nekovee, "Vehicular AD HOC and grid networks: Discussion, design and evaluation," in *Proc. 14th World Congr. Intell. Transp. Syst. (ITS)*, vol. 2. 2007, pp. 1555–1562.
- [206] L. A. Tawalbeh, A. Basalamah, R. Mehmood, and H. Tawalbeh, "Greener and smarter phones for future cities: Characterizing the impact of GPS signal strength on power consumption," *IEEE Access*, vol. 4, pp. 858–868, 2016.
- [207] L. A. Tawalbeh, R. Mehmood, E. Benkhlifa, and H. Song, "Mobile cloud computing model and big data analysis for healthcare applications," *IEEE Access*, vol. 4, pp. 6171–6180, 2016.
- [208] F. Piccialli and J. E. Jung, "Understanding customer experience diffusion on social networking services by big data analytics," *Mobile Netw. Appl.*, pp. 1–8, Dec. 2016.
- [209] M. Büscher *et al.*, "Intelligent mobility systems: Some socio-technical challenges and opportunities," in *Communications Infrastructure*. *Systems and Applications in Europe* (Lecture Notes of the Institute for Computer Sciences, Social-Informatics and Telecommunications Engineering), vol. 16. Heidelberg, Germany: Springer, 2009, pp. 140–152.
- [210] A. Chianese, F. Marulli, F. Piccialli, P. Benedusi, and J. E. Jung, "An associative engines based approach supporting collaborative analytics in the Internet of cultural things," *Future Generat. Comput. Syst.*, vol. 66, pp. 187–198, Jan. 2017.
- [211] A. Chianese and F. Piccialli, Improving User Experience of Cultural Environment Through IoT: The Beauty or the Truth Case Study. Cham, Switzerland: Springer, 2015, pp. 11–20.
- [212] S. Cuomo, P. De Michele, F. Piccialli, A. Galletti, and J. E. Jung, "IoT-based collaborative reputation system for associating visitors and artworks in a cultural scenario," *Expert Syst. Appl.*, vol. 79, pp. 101–111, Aug. 2017.
- [213] I. M. Pires, N. Garcia, and F. Flórez-Revulta, "Multi-sensor data fusion techniques for the identification of activities of daily living using mobile devices," in *Proc. ECML PKDD*, 2015, pp. 1–10.
- [214] I. M. Pires, N. M. Garcia, N. Pombo, and F. Flórez-Revuelta, "Identification of activities of daily living using sensors available in off-the-shelf mobile devices: Research and hypothesis," *Adv. Intell. Syst. Comput.*, vol. 476, pp. 121–130, May 2016.



RASHID MEHMOOD (M'06–SM'10) is currently the Research Professor of Big Data Systems and the Director of Research, Training, and Consultancy with the High Performance Computing Centre, King Abdulaziz University, Saudi Arabia. He has gained qualifications and academic work experience from universities in U.K., including Oxford, Birmingham, Swansea, and Cambridge. He has over 20 years of research experience in computational modelling and simulation systems

coupled with his expertise in high-performance computing, big data, and distributed systems. His broad research aim is to develop multi-disciplinary science and technology to enable a better quality of life and smart economy with a focus on real-time intelligence and dynamic system management. He has authored over 100 research papers, including four edited books. He has led and contributed to academia-industry collaborative projects funded by EPSRC, EU, U.K. regional funds, and Technology Strategy Board UK with the value over £50 million. He is a Founding Member of the Future Cities and Community Resilience Network. He is a member of ACM and OSA, and the former Vice-Chairman of IET Wales SW Network. He has organised and chaired international conferences and workshops, including EuropeComm 2009 and Nets4Cars 2010 to 2013.



IYAD KATIB received the B.S. degree in statistics/computer science from King Abdul Aziz University (KAU) in 1999 and the M.S. and Ph.D. degrees in computer science from the University of Missouri–Kansas City in 2011 and 2004, respectively. He is currently an Associate Professor with the Computer Science Department and the current Vice Dean and the College Council Secretary with the Faculty of Computing and Information Technology, KAU. He is also the Director of the KAU

High Performance Computing Center. His current research interest is on computer networking and high performance computing.



NASSER N. ALBOGAMI received the Ph.D. degree from U.K. He is currently an Assistant Professor with the Information Technology Department, Faculty of Computing and Information Technology, and the Vice-Dean of the Faculty of Tourism, King Abdulaziz University. His current research interests are in distance eLearning and Tourism systems.



FURQAN ALAM was born in Jaipur, India, in 1987. He received the B.C.A. degree from the Birla Institute of Technology, Ranchi, India, in 2007, the M.C.A degree from Rajasthan Technical University, Kota, India, in 2010, and the M.Sc. degree in computer science from the King Abdulaziz University, Jeddah, Saudi Arabia, in 2015, respectively, where he is currently pursuing the Ph.D. degree in computer science. He was an ABAP Programmer with DSIT solutions, Bengaluru, India.

He was a SAP ABAP Consultant with Bhaskar Industries Ltd., Bhopal, from 2011 to 2012. He is currently associated with the Faculty of Computing and Information Technology, King Abdulaziz University. His main areas of research interest are advance networking, Internet of Things, and artificial intelligence. In 2015, he was awarded by the university for his excellent performance in the first year of Ph.D.



AIIAD ALBESHRI received the M.S. and Ph.D. degrees in information technology from the Queensland University of Technology, Brisbane, Australia, in 2007 and 2013, respectively. He has been an Assistant Professor with the Computer Science Department, King Abdulaziz University, Jeddah, Saudi Arabia, since 2013. His current research focuses on security and trust in cloud computing and big data.